

Received April 29, 2020, accepted May 14, 2020, date of publication May 22, 2020, date of current version June 5, 2020. *Digital Object Identifier 10.1109/ACCESS.2020.2996584*

Factors That Influence the Adoption of Enterprise Architecture by Public Sector Organizations: An Empirical Study

NOR AZIZAH AHMAD[®], SULFEEZA MOHD. DRUS, AND HAIROLADENAN KASIM

Informatics Department, College of Computing and Informatics, Universiti Tenaga Nasional, Putrajaya 43000, Malaysia

Corresponding authors: Nor Azizah Ahmad (norazizah.ahmad@gmail.com), Sulfeeza Mohd. Drus (sulfeeza@uniten.edu.my), and Hairoladenan Kasim (hairol@uniten.edu.my)

This work was supported in part by the Energy University, Malaysia (UNITEN).

ABSTRACT The adoption of enterprise architecture (EA) is deemed to be low despite EA's substantial benefits to organizations, especially in the public sector, as evidenced by the scarce literature on the EA adoption stage, which focuses on the decision or intention to adopt EA. This research attempts to identify the significant factors that influence EA adoption in the public sector by using the technology-organizationenvironment framework and organizational theory. The conceptual model was validated using partial least squares structural equation modeling with data collected from 255 key informants from public sector organizations. Empirical results show that clear communication, coercive pressure, expected benefit, good governance, mimetic pressure, normative pressure, and organizational size have a significant influence on the adoption of EA by public sector organizations. Surprisingly, top management support, ICT infrastructure, EA complexity, organizational readiness, and external support have a nonsignificant influence on EA adoption by public sector organizations. The implications of the findings are also discussed in theoretical, contextual, and practical aspects of EA adoption in the context of public sector organizations. This model facilitates decision-makers in focusing only on the significant factors in the organizational and environmental context that influence EA adoption in Malaysian public sector organizations. Subsequently, the findings may contribute to better insights for EA initiators in their EA implementation plan and strategic planning to support EA implementation among Malaysian public sector organizations.

INDEX TERMS Adoption, enterprise architecture, institutional theory, public sector, TOE.

I. INTRODUCTION

Enterprise architecture (EA) is a strategic tool for defining the overall structure and operation of an organization. EA comprises multiple viewpoints, including business, application, information, and technology, which enable organizations to understand their current state and promote a desirable future information technology (IT) model [1]. Gartner [2] predicts that by 2021, enterprise architects will be helping organizations plan and design technology innovation toward digital services and new customer experiences. Consequently, EA benefits decision-makers and IT practitioners by achieving desired technology solutions [3], eliminating the duplication of applications and processes [4], allowing interoperability, and improving coordination between business

The associate editor coordinating the review of this manuscript and approving it for publication was Shouguang Wang^(D).

and IT [5]. Despite its substantial benefits, however, EA is not widely embraced [6].

Electronic government (e-government) is one of the substantial services of the digital government to ensure that good, seamless, and connected public sector service is provided, as clearly stated in the report of the 2010 UN e-Government Survey [7], [8]. As a result of the e-government report, the adoption of EA has become a primary strategy to improve public service delivery and contribute to national development [8]. Many developed countries such as the United States, Denmark, and Finland have proactively encouraged the implementation of EA [9]–[11]. An EA survey by.McKinsey [1] conducted among CIOs, heads of EA, and business leaders working in EA fields found that EA plays a vital role in running digital transformation [12], [13]. In their report, 62% of organizations recognize EA as an enabler of digital transformation [1].

EA was realized in the Malaysian public sector's (MPS) information communication technology (ICT) transformation model in 2013 as a transformational program and an enabler of digital transformation. Therefore, the MPS has allocated a large investment in the development of the Malaysia Government Enterprise Architecture (MyGovEA) [14]. MyGovEA, formerly known as 1GovEA, is a government EA framework that enables initiatives in the digital government agenda for the public sector to be achieved effectively. MyGovEA is clearly incorporated in the Public Sector ICT Framework of the Public Sector ICT Strategic Plan (ISP) 2016–2020 and followed by a series of workshops, awareness, and training for MPS organizations [15]. Initially, the government targeted 25 public sector agencies to have implemented EA by the end of 2016, but only five agencies adopted EA, including two federal agencies, which were included in the EA pilot project [16]. A report published by the Malaysian Administrative Modernization and Management Planning Unit (MAMPU) reveals that the EA capability maturity assessment of MPS indicated that the EA adoption of the MPS is moving toward level 2, or the formalized stage [17]. This finding proves that the EA projects of government agencies are still at a low degree and its significance has yet to be proven [14]. Thus, a mechanism or method to help the MPS adopt EA is clearly needed.

The decision-making process to institutionalize EA in the public sector organization is slow, which indicates a low success rate [18]–[20]. This condition might be related to the absence of suitable EA adoption guidelines to follow, the lack of success stories from organizations that have successfully adopted EA, or the lack of awareness of the existence of EA in the public sector [21].

Suitable EA adoption models that consist of significant factors of EA adoption is absent in the literature, as a great deal of previous EA research has focused on the post-adoption and implementation phases [22]-[25]. The current literature also lacks focus on the EA adoption phase, which is the intention to adopt EA at the organizational level [26]. A review of EA adoption literature shows that the current adoption models are applicable for the post-adoption phase, such as the assessment model by Bakar et al. [27] and the critical success factor of EA adoption by Lange et al. [22], instead of adoption issues, particularly in developing countries. Furthermore, EA adoption studies, especially in developing countries, remain scarce [20]. No scientific research on EA adoption in the MPS organization context was found at the beginning of the research project [21], [26].

To address the gaps in the existing literature and the problem with EA adoption by MPS organizations, this research investigates the factors that influence organizational adoption of EA by MPS. To achieve this objective, the following main research questions are raised: What are the factors that influence EA adoption by MPS organization? What model can be used to explain factors that influence EA adoption in MPS organizations? This research employs a quantitative approach, using survey questionnaires to answer these questions. This research followed the partial least squares structural equation modeling (PLS-SEM) research design [28]. The survey was distributed randomly to the decisionmakers of MPS organizations. This research integrates the technology–organization–environment (TOE) framework and institutional theory as underpinning theories in developing the conceptual model. This research identifies the organizational and environmental factors that influence EA adoption by MPS organizations. Our findings may assist the top management of public sector organizations in strategizing EA adoption by recognizing the important factors of EA adoption in their organization.

This paper is composed of nine themed sections. Section 1 introduces the research. Section 2 conducts a literature review regarding EA and the theoretical background. Section 3 describes the model development and hypotheses of this research. Section 4 explains the methodology on the basis of the research design that this research employed. Section 5 demonstrates the results of the analysis. Section 6 discusses the findings of the research in detail. In Section 7 elaborates on the research implications, followed by the research limitations and future work. Finally, the paper ends with the conclusion.

II. LITERATURE REVIEW

A. ENTERPRISE ARCHITECTURE

Various definitions of EA are found in this field. First, from the perspective of enterprises, architecture could be defined as a set of descriptive representations (i.e., models) that are relevant for describing an enterprise such that it can be produced according to management's requirements (quality) and maintained over the period of its useful life (change). Zachman (1996) defined EA as the architecture that is a set of design artefacts, or descriptive representations, that are relevant for describing an object such that it can be produced to requirements (quality) as well as maintained over the period of its useful life (change). References [29], [30] described EA as a coherent whole of principles, methods, and models that are used in the design and realization of an enterprise's organizational structure, business processes, information systems, and IT infrastructure. This definition is consistent with that of Gilliland et al. [31], who said that EA provides a long-term view of the organization. In addition, EA is defined as the process of translating and converting strategic requirements into processes, data, and technology, providing the organization's big picture in detail and handling change management [32]. Although various definitions of EA exist, some consensus appears to have been reached with regard to the idea that EA refers to the ISO/IEC/IEEE FDIS 42010:2011 standard definition [33] and is also stated in one of the prevalent books on EA [34]. In the present research, EA is defined as a holistic view that effectively integrates different domains in business, data, application, and information in organizations.

The non-harmonized view of EA could cause confusion among novice EA stakeholders, such as the MPS. As such, one of the main motivations of this research is to assist EA adoption in the MPS, because EA acts as a middleman between business and IT, as well as plays an essential role in facilitating the evolution of high-level capabilities at the organization level [13], [35] and managing the change from the current state to the future state [31], [36].

B. TECHNOLOGY ADOPTION MODELS

Notable theories and models that focus on the individual level of adoption include but are not restricted to the theory of reasoned action [37], the theory of planned behavior (TPB) [37], the technology acceptance model (TAM) [38] and the extended TAM [39], the unified theory of acceptance and use of technology [40], and the model combining TAM and TPB [41]. At the organization level of adoption, TOE [42], diffusion of innovation (DOI) theory [43], institutional theory [44], resource-based theory [45], and the DeLone and McLean information system success model are prevalent theories in IS research. Rogers Everett [43] proposed DOI for innovation acceptance and adoption among individuals and organizations. However, DOI focuses only on technology characteristics and does not emphasize the environmental aspect. The TOE framework posits that the decision of adoption decision is strongly influenced by three contexts, namely, technological, organizational, and environmental. The TOE framework provides generic contexts, thus allowing easy inclusion of new predictors [46].

Institutions or organizations are a critical component in the environment and exert three types of pressure, namely, coercive, normative, and mimetic [47], [48]. These pressures are present in institutional theory. Institutional theory focuses on the extensive and more robust characteristics of the social framework [49], unlike the DeLone and McLean model, which focuses on six significant dimensions, namely, information, system, and service quality, (intention to) use, user satisfaction, and net benefits [50], and is thus more suitable to IS products that are already implemented in the organization or marketplace. The integration of TOE and institutional theory is deemed suitable as both theories incorporate the multidimensional factors involved in the organizational adoption of IS or technology such as EA. Apart from one of the most popular IS models, the TOE framework has three technological, organizational and environmental contexts, while the three factors of the institutional theory are present in the organizational and environmental contexts [51]. This allows for easy integration into the TOE framework [46]. Moreover, it can be seen from the relevant literature that these two theories have not been used to examine the adoption of EA. On this premise, this research adopts the TOE framework and institutional theory on the basis of their relevance in explaining the factors of EA adoption by MPS organizations

C. TOE FRAMEWORK

The TOE framework is an organizational-level theory that consists of technological, organizational, and environmental contexts [42]. It has been described as a generic theory because it allows easy inclusion of additional constructs or factors [46]. The TOE framework is consistent with DOI theory, as DOI adoption is comparable to the elements in the TOE framework's organizational and technological contexts [51]. The TOE framework has proven useful for a wide range of innovations and contexts, and it has been broadly supported by empirical and well-established works [52]. The adoption of innovations is affected by technological, organizational, and environmental contexts within a firm [51], [53]. Therefore, many IS studies have adopted the TOE framework in various settings such as halal warehouse service [54], electronic customer relations management (CRM) [55], e-procurement [56], RFID [57], e-government [58], open government data (OGD) [59], open platform [60], cloud computing [61], software as a service (SaaS) [62], electronic record management system (ERMS) [63], and the Internet of Things (IoT) [64]. This feature strengthens the rationale for adopting the TOE framework for this research because it provides clear and important empirical support for research. On the basis of the literature review, the three contexts of the TOE framework underpin this research to identify factors that influence the organizational adoption of EA by MPS.

D. INSTITUTIONAL THEORY

Institutional theory focuses on pressures, namely, coercive, normative, and mimetic pressures [44]. Institutional theory is used to analyze diverse sectors and organization fields. Pressures suit the organizational and environmental context, because the theory probes how pressure is created, diffused, adopted, and adapted over space and time by rules and regulations, cultural expectation, and imitation from other organizations [65]. Pressures appear as the influential factor for organizational reform [66]-[68] and have been examined as motivational pressures in green IT and suggested as a motivational factor for EA adoption [9]. As stated earlier, the elements of institutional theory are comparable to those of the TOE environmental context, such as regulatory and internal pressures. Other studies analyzed pressure in different contexts, such as those on green IT by Kuo and Dick [69], ecological sustainability by Chen et al. [65], and public sectors by Hjort-Madsen [68].

From the theoretical lens, EA requires a wide-ranging theory to explain its adoption challenges in organizations. However, few studies have analyzed the determinants from the perspective of pressure in EA adoption studies [67]. Prior research on the adoption of EA revealed a dearth of environmental and organizational pressure factors. Therefore, this research deems that institutional theory is relevant in investigating EA adoption in the context of influence factors within the scope of this research.

E. RELATED WORKS

The literature review revealed that the EA adoption model has different names, such as 3D model [9], improved EA adoption method (EAAM) [19], framework for analyzing the change and influence of EA programs and their institutionalization [67], model of resistance during the EA adoption process [70], and knowledge relationship model of EA and top management roles [71]. Previous studies on EA adoption are summarized in Appendix A. Evidently, previous studies investigated and analyzed the topic from fragmented contexts, such as the context of pressure. For example, Gamiet [72] proposed a model to avoid the application duplication problem and to recognize the diversity of different EA adoption projects [9], [73] proposed an integrated EA model that focuses on social and technical impact only. EAAM was proposed to reduce resistance to the EA adoption process [19]. The EAAM model addressed the issue of lack of understanding of EA concepts [19]. However, these models are formulated based on different perspectives and issues. The literature also indicated a lack of EA adoption studies despite increasing attention on IT/IS adoption studies from developing countries.

The first study on EA in the Malaysian context investigated EA practices in Malaysian enterprises, including public and private sectors [74]. Business and IT alignment were found to be the most important business issue in EA, followed by business change and application renewal [75]. The study by [78] revealed that the issues and challenges faced by MPS organizations include the lack of EA awareness and readiness, limited EA knowledge and skill among team members, improper governance, absence of a mandate from the stakeholders or the government, and the absence of an EA tool to maintain EA artefacts. Therefore, their study identified the factors that are relevant to EA adoption in the context of MPS. Other EA studies have been performed in the context of the MPS on the EA implementation model [76] and the EA readiness assessment model [77]. However, these studies have a narrow focus, concentrating only on talent management, readiness, and the implementation phase.

Although the TOE framework has been adopted in various IS research, it has not been extensively applied to the domain of EA, particularly in the area of EA adoption. EA has been the topic of a variety of empirical research with specific purposes and findings. Such research also investigated various factors that influence the adoption of EA. To date, the adoption and implementation of the EA in developing nations have been minimal [20]. Countries face difficulties in adopting the EA; the challenges are similar and do not rely on the nature of the country. Studies on the public sector's adoption of EA are fewer than those on other sectors, such as small and medium-size enterprises and the private sector. Nevertheless, the experience, culture, and purpose of the sectors in implementing EA are different. The perceptions gained in the various sectors are not exchangeable in terms of the design, model, and viewpoints of organizations.



FIGURE 1. MyGovEA framework.

F. EA IMPLEMENTATION IN THE MALAYSIAN PUBLIC SECTOR

In 2014, MAMPU introduced MyGovEA as the national EA framework. The development of EA was based on The Open Group Architecture Framework. MyGovEA provides ecosystem services for the public sector. MPS also realized that the benefit of EA toward digital transformation in government services could be accomplished by adopting EA [78]. Further acceleration of the use of MyGovEA is specifically stated in the MPS Strategic Plan 2016-2020 [15]. The government targeted 25 government agencies to implement EA in designing desirable IT solutions. MAMPU reported that sustaining EA practices in public sector agencies is challenging [79]. Hence, EA in MPS is still in its infancy. The survey conducted by MAMPU in 2014 on EA capability maturity assessment reveals that the MPS is moving toward level 2 (formalized stage) with regard to the adoption of EA practices. The survey also examined the EA framework used by MPS organizations.

MyGovEA is guided by the developed architectural framework to define the core elements of the architecture in developing and implementing EA practices. As shown in Figure 1, the MyGovEA framework acts as a guide to describe the main components of MyGovEA. The components of MyGovEA are vision, governance, architecture domains, principles, tools and repository, and methodology. MPS organizations have to develop these components because they are fundamental to the implementation of the EA practice.

III. MODEL DEVELOPMENT AND HYPOTHESES

A review of IS/IT technology adoption and implementation suggests that many technological, organizational, and environmental factors are likely to affect the adoption of EA. Adoption is frequently observed as a dependent variable in the related literature [80]. Independent variables are categorized in the three contexts of TOE: 1) technological, 2) organizational, and 3) environmental. This research adopts two technology adoption models, namely, the TOE framework and institutional theory, which have been widely adopted for



IS studies in the organizational context. The factors were also found from prior research on EA and TOE conducted across different industries in different countries, as shown in Appendix A. This research proposes 12 TOE factors that could influence EA adoption by public sectors, as illustrated in Figure 2: sufficient ICT infrastructure, EA complexity, top management support, organizational readiness, clear communication, normative pressure, expected benefit, good governance, organization size, external support, mimetic pressure, and coercive pressure. These factors are the most utilized factors for the three principle contexts of the TOE framework and institutional theory. This research also proposed clear communication as a specific factor of EA adoption phases within the organizational context of the organization. The selection of these twelve factors was based on the mapping analysis of the systematic literature review of the adoption models in the EA studies [21], [26] and the prior TOE-based technology adoption studies (see Appendix A). Factors that have appeared from both these domains and thus are believed to be essential for understanding and explaining the adoption of EAs have been proposed in the research model. Therefore, the researcher measured these factors to explain the intention or decision of an MPS organization to adopt EA. The researcher formulated the following hypotheses to assess the proposed model of EA adoption by MPS organizations.

A. TECHNOLOGICAL FACTORS

The technological context includes internal and external technologies that are relevant to the organization. The technologies may include types of equipment and processes that can foster EA adoption, as found by Depietro *et al.* [42]. Many studies argued that technology plays an important role and found a positive effect on innovation adoption at the organization level [54], [55], [59], [60], [62], [81]. The accessible technology variables are perceived advantage, perceived easiness, compatibility, observability, trialability, compatibility, complexity, and perceived barrier. However, this research employs the technology variables of sufficient ICT infrastructure and EA complexity, as suggested by [9], [13], [19], [82]–[85]. The rationale of measuring technology readiness, which is the degree to which the organization has the necessary technology infrastructure and IT human resources to implement the EA, is crucial to the pace of adoption because organizations that have the necessary technology will be able to adopt EA faster than organizations that do not have the technology. This variable is also found to be significant by the study of Interorganizational Business Process Standards [86].

EA complexity has been recognized as a technological factor because it affects time, cost, and management control [87], [88]. Technological complexity is the degree to which the use of technology is free of effort [89]. Analyses of the relationship between technological factors and technology adoption can be extended to EA adoption. Therefore, this research proposes the following hypotheses:

H1: Sufficient ICT infrastructure positively influences the intention to adopt EA.

H2: EA complexity significant positively influences the intention to adopt EA.

B. ORGANIZATIONAL FACTOR

The organizational context refers to the characteristics, internal pressure, and resources of the organization, including top management support, organizational readiness, clear communication, normative pressure, expected benefit, good governance, and organization size, which are among the most accepted predictors of innovation adoption [53], [90]. These characteristics also refer to the descriptive measure. However, the extent to which these organizational variables influence the phases of adoption has not been examined [53]. Top management support has been considered one of the most influential organizational factors for IT adoption in organizations [91], [92]. Other scholars argued that top management support is accountable for the norms, cultures, values, visions, and missions, e.g., Balaid et al. [93], which eventually encompass the entire community in the form of regulations, policies, routines, and procedures, and serve as powerful templates. Wang et al. [94] revealed that top management support provides the necessary involvement, resources, and authority in guiding and assisting innovation. For example, financial resource has long been posited as a barrier to innovation adoption [95].

In contrast to the other studies that focus on the postadoption stages, including those that focus on the actual use of innovation as in e-business, such as the study by Zhu and Kraemer [80], this research focuses on the adoption stage (or intention to adopt). EA involves costly investment in hardware, software, system integration, and change management [31]. Therefore, sufficient financial resources help the organization obtain these necessary resources and develop EA skillsets and competency. The researcher argued that in the attempt to make innovation more cost-effective for the organization, a large amount of money needs to be allocated for innovation, as it will increase the motivation to innovate more seriously and actively within the organization [96]. This finding is consistent with that of other scholars who studied EA [13], [19], [82], [83]. The choice of this variable also emerged during an interview session with the EA practitioner team. Another variable, which is top management support, is also deemed important in the adoption of innovation and has been found to positively affect RFID, SaaS, e-government adoption, and cloud computing [58], [62], [97]. Specific studies on EA [9], [68], [82]–[84], [98]–[100] found that top management support is a critical determinant of adoption. In this regard, the researcher hypothesizes the following:

H3: Top management support positively influences the intention to adopt EA.

Organizational readiness is another crucial variable in innovation adoption [94], [101]. Organizational readiness includes support from different organizational levels, adequate technical support, experienced people, and EA knowledge and skill within the organization, and it can provide a significant business advantage [102]. In general, EA knowledge and skill represent the totality of organizational EA knowledge and skilled personnel within an organization. This factor is urgently required among organizational employees for adoption intention [10], [19], [84], [103]. In the context of EA adoption, EA knowledge and skill are practical ways in which organizations can promote EA adoption. In return, the organization would benefit from the return on investment [104]. Hence, organizational readiness is considered a driver of adoption in similar IS studies, such as those on halal warehouse service by Ngah et al. [54], IoT by Hsu and Yeh [64], and OGD by Wang and Lo [59]. The present research hypothesizes the following:

H4: Organizational readiness positively influences the intention to adopt EA.

An additional factor is recognized under organizational context, namely, clear communication. Clear communication constitutes another important variable in the EA literature [83], [84]. and prevalent studies discovered that one of the problems hindering EA adoption is communication failure between the EA team and the business and IT personnel [9], [84], [105]. To understand the value of EA, communication among different stakeholders is required. Communication can give positive perceptions and identify the potential values of EA [106]. This variable is strongly recommended by.Hjort-Madsen [98], who said that IT planning must address the language gap between business and IT personnel to gain a mutual understanding of the organization's strategies and objectives. In this case, team members in the organization feel that they can state their opinions, thoughts, and feelings without fear. This variable appears vital in the TOE framework and DOI [42], [43], as both are complementary. Therefore, the next hypothesis is proposed as follows:

H5: Clear communication positively influences the intention to adopt EA.

Normative pressure is "driven by pressures brought about by professions. One mode is the legitimization inherent in the licensing and crediting of educational achievement. The other is the inter-organizational networks that span organizations. Norms developed during education are entered into organizations" [48]. According to DiMaggio and Powell, the norms will develop through education within the organization. As a result, people from the same educational backgrounds will approach problems in much the same way. In this case, organizational culture holds the uniqueness in every organization. The study by [110] reveals that the role of organizational culture is significant so that the amount of needed investment can be known given limited resources for EA management in different organizational cultures [107]. The organization has a different perspective and reaction toward the intention to use or adopt EA and hence leads the organization to change, which is one of the most difficult things to achieve [9], [107]. Organizational culture is portrayed in terms of how committed employees are to the common objectives and decisions in EA adoption. Therefore, normative pressures result from the demands of professional associates, organizational culture, and the extent to which the government promotes the use of information technology and especially EA [108], [109]. This variable has been shown in a study by Weiss [110] to significantly influence EA and is suggested for EA adoption [9], [10], [19], [98]. Hence, the present research hypothesizes the following:

H6: Normative pressure positively influences the intention to adopt EA.

Including the expected benefit of adopting EA for the organization as one of the factors that influence EA adoption is also reasonable. Return on investment is a consideration in weighing the expected benefit before the decision to adopt EA is made [111]. Expected benefit can be seen as the relative advantage, "who saw it as the degree to which an innovation is perceived as providing greater organizational benefits than the idea it supersedes or the status quo." Rogers Everett [43] perceived expected benefit as a part of the organizational context. For example, an IoT adoption study found that expected benefit is an influential factor in Taiwan's logistics industry [64] and the mobile supply chain management system in manufacturing firms in Malaysia [111]. EA benefit is an important factor that influences an organization to adopt EA [9], [13], [18], [19], [30], [105]. Therefore, the next hypothesis is proposed as follows:

H7: Expected benefit positively influences the intention to adopt EA.

Good governance includes the strategy and the operating model in terms of defining the roles, responsibilities, and procedures used by an organization for internal activities and EA adoption [112]. Although governance has a significant adverse effect on the study of e-participation and e-government maturity from a global perspective (Krishnan *et al.* [113]), it is highly suggested by other scholars [9], [13], [18], [19], [82]–[85] for EA adoption. To institutionalize EA in the organization, a new governance regime must be introduced [114]. For example, good governance is significantly found as an indicator between ICT and

socioeconomic development in developing countries because the concept of governance is gaining increasing focus as a national-level construct [115]. According to [115], "Governance is thus responsible for creating an environment that enables the participants in all aspects of the economy to easily evolve, learn and adapt while being publicly and openly accountable." Other scholars mentioned that governance is not the same as government, describing it instead as "the action of the state and, besides, encompasses actors such as communities, businesses, and NGOs." [116]. EA allows interconnection of governments through online services [8], [117], thus providing supervision or a control communication network from within the industry, especially in the public sector for governance. Previous scholars claimed that through such a network, all the diverse interest groups from within the public sector can interact and reach consensus on important issues that are beyond the unitary control of one particular stakeholder [115], [118], [119]. The role of governance is an antecedent in shaping the economic and social development of a country. However, the issue of governance has rarely been raised in mainstream IS research, especially when considering macro-level indicators [115], such as an MPS organization. For example, EA documents evolve as business processes change. Thus, the government should keep up with this agility [106]. Therefore, the variable is considered to be measured in this research, and the following is posited:

H8: Good governance positively influences the intention to adopt EA.

Organization size refers to the organization's number of employees [120]. Previous studies by Seppänen [9], Korhonen and Halen [13], Syynimaa [19], Shaanika and Iyamu [84] reveal that organization size critically influences the adoption of EA. Other specific studies confirm that size is the critical factor for technology, e-procurement, and RFID adoption, although it is not critical in techno-relationship innovation [56], [57], [108]. The inclusion of this variable in this research is important and has a possible effect on EA adoption, as indicated by literature reviews and the TOE framework [42]. Other prominent scholars [42], [80], [101] found that a large organization makes an EA adoption decision more quickly than smaller organizations do because they have a greater need to stay at the leading edge of technology. These scholars also mentioned that larger organizations have benefited because they have more resources for investing in and adopting innovation. Therefore, this research proposes the following hypothesis on the basis of the evidence:

H9: Organization size positively influences the intention to adopt EA.

C. ENVIRONMENTAL FACTOR

The environmental context refers to the sector within which the organization operates, the external pressures, and the opportunities that may influence the EA adoption [46]. Concerns are given to three critical environmental factors that are factored into this conceptual research model; they are external support, coercive pressure, and mimetic pressure, as presented in Figure 2. External support refers to the existence of vendors, agencies, and businesses in the external environment of the organization to support the adoption of EA activities [121], that is, a third-party dependency in which the group relies on IT suppliers for problem determination and resolution, customization, interfaces, and functional enhancement to new IT [122]. EA is a long-term view of the organization, according to Gilliland *et al.* [31], and may therefore be a lifelong commitment for many organizations. External support was identified and recognized in the technology adoption phase especially in enterprise resource planning (ERP) [123]. According to Baker [46], support from the availability of skilled labor, consultants, or third-party suppliers of a technology is important to foster the new idea being adopted in the organization. Consequently, external support is a vital factor in procuring, installing, maintaining, and training regardless of the type of technology that the organization has. Therefore, this research proposes the following hypothesis on the basis of the evidence:

H10: External support positively influences the intention to adopt EA.

Pressure focuses on external pressures, which are coercive and mimetic [48]. These pressures are presented in institutional theory, which has been widely used in IS research to understand the mechanisms of adoption and implementation of innovation in organizations. This theory postulates that organizations are influenced by external pressures when forming organizational structures [48]. Such pressures have been shown to significantly affect adoption in IS studies [53], [124], [125]. Furthermore, previous studies revealed that institutional theory is relevant when examining OGD, as Wang and Lo [59]; assimilation processes; and technology adoption [108]. Thus, this research forecasts that such pressures remain a significant influence on EA adoption in the organization, as described in institutional theory.

Coercive pressure refers to "the formal pressure and external pressure exerted upon them by other organizations upon which they are dependent, and the cultural expectations in the society within which the organization's function" [48]. From previous EA adoptions, such pressure suggests that regulation could affect project results in negative and positive ways [10]. Furthermore, this pressure has a great influence during the adoption phase by force of mandate [11]. This variable is selected because its potential effect on EA was identified in previous studies [9], [10], [66], [99], [126]. This finding is consistent with the study of Pudjianto *et al.* [58], who found that lack of a supportive regulatory environment for e-government will result in a negative effect on assimilation. Therefore, the following is posited:

H11: Coercive pressure positively influences the intention to adopt EA.

Previous studies maintain that mimetic pressure is more noticeable at the early stage of innovation diffusion, where the uncertainty of outcome is high [60], [127]. They stated that few organizations adopt innovation at the early stage in belief of its efficiency. This action then influences other organizations that have not adopted the innovation to surrender to pressure because they are worried about appearing different. Such capitulation then generates added bandwagon pressure [128]. According to Shim *et al.* [60], this pressure induces other organizations to follow the decision to adopt an innovation. Many reviews of this notion have been conducted [44], [129], [130]. Hence, this research expects mimetic pressure to have a salient influence at an early stage. Therefore, the present research hypothesizes the following:

H12: Mimetic pressure positively influences the intention to adopt EA.

IV. METHODOLOGY

The operational research design outlines the sequence of the phase of the research process involved in this research. The process of this research follows the Urbach [28] model, which is typically applied in SEM-based research. The operational research design of this research comprises four steps: conceptual model development, instrument development, data collection, and model validation. First, the conceptual model was developed based on comprehensive literature reviews of EA adoption and IT/IS adoption studies. Twelve factors from three perspectives influence EA adoption by MPS organizations. The technological factors consist of sufficient ICT infrastructures and EA complexity, while organizational factors include top management support, organizational readiness, clear communication, normative pressure, expected benefits, good governance, and organization size. The environmental perspective consists of three factors: external support, coercive pressure, and mimetic pressure.

Second, a questionnaire was developed as a data collection instrument. In the survey, the measurement items were adapted from past studies and were revised in terms of content validity based on expert reviews. Experts from the academic and industry sectors were identified based on their expertise in various fields, such as IS, EA, research methods, statistics, academic, and public sectors. The experts' reviews found that two items were rejected and only 50 items remained. Subsequently, a pilot test was conducted to assure the reliability and the quality of the questionnaire. Third, the survey was distributed randomly to actual respondents in MPS organizations. The unit of analysis of this research is the organization. Therefore, the survey was sent to the chief information officer (CIO) or head of the organization, such as mayor, director general, or manager. These respondents were chosen because they are involved in the decision-making process of the organization. A total of 255 valid responses were collected from the survey. Finally, these responses were analyzed to validate the model by using PLS-SEM, which involves two types of analysis: measurement model and structural model analyses.

A. INSTRUMENT DEVELOPMENT

The instrument development takes into account the design of the questionnaires. This research adapted the instrument development procedures in [131]. The validity of the

98854

questionnaire was determined by conducting a pre-test and pilot test to ensure the appropriateness and sensitivity of the questions. The second milestone was the completion of data collection, which was achieved by determining the sampling procedures and administering the survey. Then, on the basis of the collected data, the causal relationship outlined in the conceptual model was analyzed in the next phase.

The instrument that was used in this research is a questionnaire. Therefore, the measurement items were collected using a combination of deductive and inductive approaches. The researcher reviewed literature published in high-impact journals such as MIS Quarterly, showed empirical information such as Cronbach's alpha, and is widely cited and used in relevant research [132]. Multiple items were gathered in the item pool to measure the different dimensions of the construct. Then, the items to be measured were adapted from previous relevant research. The instruments that were widely cited and frequently used by other researchers are good criteria for choosing a good instrument [132]. Therefore, the content validity was fulfilled [133]. The reliability and accuracy of scores from instruments were obtained through internal consistency across the items [132]. The Cronbach's alpha is used to test the internal consistency [134]. The measurement items then underwent pilot testing. All measurement items used in this research are shown in Appendix B.

Various opinions have been expressed about the number of points on a Likert scale, such as whether the scale should be even-numbered (2, 4, 6, 8, or 10) or odd-numbered (3, 5, 7, or 9) scale [135]. A seven-point Likert scale is suggested in this research to represent the responses of the respondents [136]. The seven-point Likert scale has been proven to have the highest magnitude of factor loadings and composite reliability of the latent variables. This condition indicates a stronger association between measurement items and latent variables and demonstrates adequate discriminant validity of the latent variables [137]. Miller [138] stated that human perceptions cannot discern more than seven different categories. In addition, the odd-numbered Likert-type scale was used in this research to overcome the problem of too many neutral responses, which are common among Asian people, especially in the Malaysian context, when given the option to choose [139].

Two techniques are used to control common method biases in the research, namely, 1) procedural remedies and/or 2) statistical remedies [140]. These two techniques are adopted for this research. The procedural remedies include improving scale items through the development of items. The development of items was carefully designed to reduce the ambiguity of items in the comprehension stage of the response process. The researcher kept the questions simple, concise, and specific, and avoided double-barreled questions. Pre-test and pilot test were conducted to ensure that the questionnaire items are simple and can be clearly understood by the respondents. The researcher also included instructions for completing the questionnaire, and the terms used in the questionnaire were clearly defined at the beginning. The instructions also

 TABLE 1. Cronbach's alpha values for the pilot test.

	Variat	ole	Items	Alpha	Interpretation
Sufficient	Sufficient ICT infrastructure		3	0.931	Excellent
(ICT)					
EA complex	ity (CF	YX)	4	0.938	Excellent
Top manage	ment s	upport (TMS)	4	0.932	Excellent
Organization	nal read	liness (OR)	4	0.785	Good
Clear comm	unicati	on (COMM)	6	0.962	Excellent
Normative p	ressure	e (NP)	3	0.913	Excellent
Expected be	nefits (EB)	5	0.991	Excellent
Good govern	nance (GVR)	6	0.958	Excellent
External IS :	support	(EXT)	5	0.899	Very Good
Coercive pressure (CP)			3	0.936	Excellent
Mimetic pressure (MP)			3	0.961	Excellent
Intention to	adopt (INT)	3	0.987	Excellent

stated that the respondent's answers are anonymous and that there are no right and wrong answers. The respondents were advised to answer the questions as honestly as possible. This procedural remedy can protect respondents' anonymity and reduce evaluation hesitation.

1) PILOT TEST

The pilot test evaluates and determines the survey instrument empirically with a small sample rather than the actual sample size. The respondents of the pilot test have similar characteristics as the actual population [141], [142]. Twenty-three respondents from different MPS organization types answered the bilingual survey for this research. Statistical analysis was used to validate and verify the pilot test responses by using SPSS version 20. This research adopted the rule of thumb of Field [143] and George and Mallery [144] that a Cronbach's alpha of more than 0.7 is acceptable and reliable, which is also consistent with the idea of Kline [145] and Sekaran and Bougie [141], [146] that a value of more than 0.7 and near 1.0 is considered excellent, realistic, and reliable. With the inconsistency in determining the minimum value of Cronbach's alpha, the cut-off point for this research is 0.7, as suggested by Hair et al. [147]. Thus, no item was dropped for this research. Table 1 shows the Cronbach's alpha value of each construct. The Cronbach's alpha value for the questionnaire is 0.976.

B. SAMPLE AND DATA COLLECTION

Questionnaires were used as the data collection instrument and were sent to the heads of organizations or key respondents such as the CIO of the selected public sector organizations under study. The personnel were chosen because they have high positions and are normally involved in decision-making, aside from knowing the overall aspects and performance of the organization. The unit of analysis of this research is the organization. Therefore, the key respondents from five MPS organization types were asked to participate in the survey. The population of MPS organizations was represented by multiple strata. Thus, this research employed probabilistic stratified random sampling approach due to its efficiency in avoiding misrepresentation of population members [132], [141]. Determination of sample size was calculated using G*Power tool, a power analysis method. The prior power analysis of this research used the error probability of $\alpha = 0.05$, the power $(1 - \beta) = 0.8$, effect size = 0.15. Therefore, this research found a minimum total sample of 178 to be drawn from a population of 741.

Questionnaires were administered through online and postal approaches, and they were also distributed by hand in public sector organizations located in Putrajaya. Although better and fast response rates could be attained by hand distribution, time and cost limitations were major constraints. The public sector organizations in the sample were located all over Malaysia, and the researcher could not distribute the questionnaires by hand given the limited time and budget. The total number of organizations involved in the survey is 600. Initially, the online survey was sent to the key respondents of organizations such as the CIO, mayor, or head of the agency by email. However, the response rate was low (20%). Following Creswell [132], [141] suggestions, the researcher conducted follow-ups via email reminder, telephone call, and paper-based approach. As a result, the number of responses increased to 285. The response rate of collected data for this research is 47.5%. The benchmark of response rate for organizational research studies is within the range of 35% to 40% [148]. Therefore, the response rate of 47.5% can be reasonably accepted in organizational research studies and adequate to generalize the population of the research context.

V. RESULT

A. DATA PREPARATION

The essential step in PLS-SEM before conducting data analysis is data preparation. Four activities are involved in this step: response rate analysis, data cleaning, nonresponse bias test, common method bias test, and normality test. This research collected empirical data by using questionnaires. Therefore, data collection issues, such as missing data, suspicious response pattern, and outliers, have to be addressed and examined accordingly. Data preparation was conducted to check and solve the issues. These issues were addressed by performing four activities: response rate analysis, data cleaning, common method bias test, and normality test. The missing data on a questionnaire regarding the blank responses of the demographic profile of the respondents and the Likert rating scale questions for the items to be measured, however, did not exceed 15% and are less than 5% values per indicator [147].

The researcher also examined for suspicious response patterns. Similar or repetitive answers for numerous questions affect the data analysis. This type of response pattern is often described as straight lining. The data can be detected by the value of standard deviation = 0 among the given questions. Nine responses were affected and were therefore dropped from the dataset, as suggested by [141], [147]. An outlier can be identified by using multivariate graphs and statistics, for example, box plots. Outliers occur when combinations of variable values are particularly rare. In this research, the outlier or extreme case is identified as 1.5 times or 3 times the interquartile range below the lower quartile or above the upper quartile [147]. Twenty responses were identified as outliers and were therefore dropped from the dataset to prevent them from causing problems with coefficient correlation [149]. Therefore, 255 out of the total collected responses remained as valid responses and are considered to be accepted and adequate sample size.

The researcher checked for nonresponse bias issue using an independent T-test sample. Nonresponse bias can be analyzed by validating that both early and late respondents are not significantly different[28]. From 255 valid responses, 144 were considered as early responses and the remaining 111 were late responses. The research conducted the independent sample T-test using SPSS for early responses versus late responses on the Likert scale answers of the survey questions. The non-response bias results indicate that there is no significant difference (p > 0.05, two-tailed test) between early and late responses on any of the survey items. It can, therefore, be concluded that this research is free from non-response bias, which is a potential issue in survey-based studies.

Common method bias, also known as common method variance (CMV), identifies the error variance that is attributable to the measurement method and causes a potential problem in behavioral research. Potential CMVs were found in this research because the research method was conducted with independent and dependent variables from the same respondents [140]. According to Podsakoff [140], method biases are a problem because they cause one of the measurement errors that threaten the validity of conclusions regarding the relationships between variables. This argument is consistent with Bagozzi *et al.* [150], who noted that one source of measurement error is method variance, which may arise from a variety of sources, such as the content of specific items, scale type, response format, and the general context.

Finding a procedural remedy that meets all needs is difficult. Therefore, statistical remedies are available to control common method biases. Common statistical remedies are Harman's single-factor test, Lindell and Whitney's (2001) marker variable, collinearity, and partial correlation method [140]. The present research chose Harman's single-factor test and correlation matrix to examine the common method bias. These two tests are highly recommended for the PLS-SEM research context [28], [151]. The result of Harman's single-factor test from SPSS showed that the maximum covariance explained by one factor is 43.1%, which is less than the majority or 50% of the variances in the variable. Therefore, common method bias did not occur in this research [152]. The total variance of this research is explained by Harman's single-factor test. The result of the correlation matrix obtained by using SPSS showed that all construct values were significant (p < 0.01 and p < 0.05) which are lower than the cut-off value of r < 0.90 [150], [153]. The result of the correlation matrix shows that no correlations went beyond 0.9 to indicate a high correlation. The two statistical tests indicate that common method bias is not a problem for this research.

A skewness and kurtosis value of 0 show a perfectly normal data distribution. As suggested by Hair *et al.* [154] and Ramayah *et al.* [155], this study used an online statistical power analysis tool called Webpower (https://webpower. psychstat.org/models/kurtosis) to test the normality of multivariate data. Mardia's multivariate skewness value $\beta = 32.32$ and p < 0.01, and Mardia's multivariate kurtosis value $\beta = 260.51$ and p < 0.01, thereby showing that the distribution of data is not normal [156]. Therefore, the PLS-SEM analysis can be performed on both normal and not normal data.

B. DESCRIPTIVE ANALYSIS

Descriptive statistics was performed to analyze the demographic profile of the respondents. Afterwards, inferential statistics, which involved a two-stage approach that consists of measurement model analysis and structural model analysis, was performed. Table 2 presents the details of the demographic profile of the respondents. The respondents were classified according to distinct demographic categories, namely, organization type (OT), age, education level (EDU), position grade (GR), years of work experience (YOW), number of employees (NOE), and number of IT employees (NoIT).

The survey results demonstrate that the majority of the respondents work at the federal level (36.5%). Most of the respondents are within the range of 36 to 40 years of age (29.4%), and only 2.4% of respondents are below 26 years of age. This result indicates that respondents are generally from the young generation. Thus, technology should not be an unfamiliar topic for them. Over half of the respondents reported that their educational qualifications are a degree (56.9%), master's degree (26.7%), and doctor of philosophy (2.7%). These results indicate that the majority of the respondents come from a strong socioeconomic background and are educated. Consequently, they are capable and eligible to contribute to the decision-making for their organization. In response to position grade, 46.3% of respondents are between the grade of 41-44 and only 20% of the respondents hold a grade lower than 41. This table evidently shows that more than two-thirds of the respondents hold a position grade from 41 and above and indicates that the respondents are officers, middle management, and top management. Therefore, the majority of respondents are part of the decision-making team. The respondents that hold a position grade below 41 belong to the support group.

Notably, more than two-thirds (68.6%) of the respondents have work experience of more than 10 years in the public sector. Only 6.3% of the respondents have less than two years of work experience in the public sector. The result indicates that the respondents have adequate experience and knowledge in public sector initiatives. They can also be considered experts in undertaking specific tasks for the organization. The result

TABLE 2.	Demographic	analysis.
----------	-------------	-----------

Demographic	Category	Frequency	Valid (%)
Organization	Federal Level	93	36.5
type (OT)	Federal statutory	30	11.8
	body		
	State level	61	23.9
	State statutory body	31	12.2
	Local authority	40	15.7
Age	Less than 26 years	6	2.4
	26-30 years	16	6.3
	31-35 years	43	16.9
	36-40 years	75	29.4
	41-45 years	60	23.5
	46-50 years	23	9.0
	Above 50 years	32	12.5
Education level	Diploma	35	13.7
(EDU)	Degree	145	56.9
	Master	68	26.7
	Doctor of philosophy	7	2.7
Position grade	< 41	51	20.0
(GR)	41-44	118	46.3
	48	51	20.0
	52	24	9.4
	54	7	2.7
	JUSA/TURUS	4	1.6
Years of work	Less than 2 years	16	6.3
experience	2–4 years	10	3.9
(YOW)	5–7 years	15	5.9
	8–10 years	39	15.3
	More than 10 years	175	68.6
Number of	< 100	56	22.0
employees	100-399	81	31.8
(NOE)	400-699	38	14.9
	700–999	14	5.5
	>=1000	66	25.9
Number of IT	Less than 3	67	26.3
employees	4-10	65	25.5
(NoIT)	11-20	35	13.7
	21-30	26	10.2
	31-40	13	5.1
	41-50	12	4.7
	More than 50	37	14.5

also showed that from the 255 questionnaires, the data on the number of employees (a measurement of organization size) demonstrated that the majority of the organizations were categorized as 100–399 (31.85%), followed by more than 1000 (25.9%), lower than 100 (22%), and 400–699 (14.9%). Only a small number was classified in the 700–999 range (10.3%).

With regard to the number of employees in the IT department to which the respondents were attached, organizations with fewer than 3 IT employees were the largest group (26.3%), followed closely by organizations with 4–10 IT employees (25.5%), more than 50 IT employees (14.5%), 11–20 IT employees (10.69%), 21–30 IT employees (10.2%), 31–40 IT employees (5.1%), and 41–50 IT employees (4.7%). This finding shows that most of the organization have IT strengths, although they did not have many IT employees. The most surprising aspect of the data is in the response to the organization type. The respondents' numbers exceed the stratified sampling rule of 30 for every stratum [157]: federal level (63%), federal statutory body

(11.8%), state level (23.9%), state statutory body (12.2%), and local authority (15.7%).

C. MEASUREMENT MODEL ANALYSIS

This research assessed the reliability and validity of the reflective measurement model. The metrics used for the reflective measurement model were internal consistency reliability, indicator reliability, convergent validity, and discriminant validity.

1) INTERNAL CONSISTENCY RELIABILITY

As suggested by Hair et al. [156], the first metric to be evaluated is typically internal consistency reliability. Cronbach's alpha is a traditional criterion to estimate the reliability of intercorrelations of the indicator or item of constructs. The idea of Cronbach's alpha is that all indicators are equally reliable on the constructs, hence reflecting a relatively low reliability value. Given this limitation of Cronbach's alpha, an alternative measure called composite reliability (CR) was applied in this research. CR is a criterion to estimate different outer loadings of indicators of the constructs. The CR varies from 0 to 1, with higher values indicating higher reliability. According to Hair et al. [147], the measure of analyzing and assessing the internal consistency reliability lies between the Cronbach's alpha and CR. Therefore, this research considers and reports both reliability measures. Specifically, CR and Cronbach's alpha values of more than 0.70 are acceptable, while a value below 0.60 indicates a lack of internal consistency reliability [156]. For this research, the values of CR and Cronbach's alpha for each construct were more than 0.70. Therefore, all indicators are reliable for measuring the constructs.

2) INDICATOR RELIABILITY

Indicator reliability refers to the consistency of indicators with what it anticipates to measure. The size of the outer loading is also called indicator reliability, which reflects the proportion of the indicator variance described by the latent variable [28]. The acceptable value of outer loadings of more than 0.70 shows that the associated indicators of constructs have much in common or are consistent. Outer loading values below 0.5 should be considered for removal [156]. The path weighting scheme was selected for the inner weights' calculation. This weighting scheme provides the highest R2 value for endogenous latent variables and is commonly appropriate for all types of PLS path model specifications and estimations [158]. As shown in Table 4, the construct OR4 showed a value of 0.465, indicating weakness and that it should be considered for removal from the scale [159]. Other constructs are retained because the values of outer loadings were more than 0.70. Therefore, the indicator reliability (outer loadings) for this research was achieved.

3) CONVERGENT VALIDITY

Another metric of the reflective measurement model is convergent validity. Convergent validity is usually determined by

Construct	COMM	СР	СРХ	EB	EXT	GVR	ICT	INT	MP	NP	OR	SIZE	TMS
COMM													
СР	0.345												
CPX	0.093	0.059											
EB	0.509	0.550	0.218										
EXT	0.450	0.594	0.248	0.513									
GVR	0.594	0.610	0.221	0.875	0.531								
ICT	0.485	0.466	0.313	0.566	0.471	0.658							
INT	0.328	0.733	0.152	0.699	0.474	0.718	0.514						
MP	0.436	0.686	0.223	0.578	0.649	0.616	0.559	0.634					
NP	0.523	0.673	0.201	0.859	0.560	0.796	0.609	0.736	0.643				
OR	0.502	0.687	0.305	0.708	0.683	0.718	0.689	0.654	0.704	0.798			
SIZE	0.032	0.039	0.026	0.041	0.030	0.039	0.062	0.035	0.079	0.035	0.067		
TMS	0.438	0.533	0.218	0.566	0.353	0.570	0.519	0.523	0.490	0.673	0.805	0.136	

TABLE 3. HTMT result.

assessing the average variance extracted (AVE) and means that the construct contains more than 50% of the indicator variance. AVE is the degree to which a construct explains the inconsistency of its indicators. The result of outer loadings contributes to the AVE value. The convergent validity of the construct can be adequately achieved if the value of AVE for each construct is equal to or greater than 0.50 [150], [160]. The AVE values for every construct in this research were more than 0.50 (ranging between 0.594 and 0.943). Specifically, after one item was deleted, which is organizational readiness, the value of AVE and CR increased to 0.594 and 0.746, respectively. The results from both indicator reliability (outer loadings) and AVE values in this research demonstrated that the measurement model conforms to the convergent validity assessment.

4) DISCRIMINANT VALIDITY

After convergent validity is confirmed to not be a problem for this research, the next step was to verify the discriminant validity. Discriminant validity refers to the construct under investigation being distinct or different from one another. This metric examines the bivariate correlation of overlapping. Following Hair *et al.* [156]'s guidelines in analyzing and reporting discriminant validity by using the PLS-SEM approach, three methods are available for assessing reflective indicators, namely, Fornell and Lacker criterion, cross-loadings, and heterotrait–monotrait ratio of correlations (HTMT). Each criterion is explained below.

5) FORNELL AND LACKER CRITERION

The bivariate correlation matrix using Fornell and Larcker criterion was used to assess the correlation between constructs. Results indicate discriminant validity between all constructs. The square root of AVE (in yellow color) on the diagonal are higher than the off-diagonal's values on the respective column and row. Thus, the requirements set by Fornell and Larcker [160] were met in this condition.

6) CROSS-LOADINGS

The result of cross-loading shows that the majority of indicators loaded highly on their respective constructs. However, the indicator OR4 showed a value of 0.465, thereby showing that this item has an issue of high loading. Therefore, OR4 was deleted before structural model analysis was continued.

7) HTMT CRITERION

The third criterion for assessing the discriminant validity is the HTMT criterion of correlation. The HTMT criterion is meant for reflective indicators and can be applied for both multi-item and single-item constructs [161]. A recent study suggested that HTMT produces a more accurate and established assessment than typical criteria such as the Fornell and Larcker criterion [147]. As shown in Table 3, all construct values of HTMT fulfilled the HTMT criterion of HTMT_{0.90} [162]. The result ascertains the discriminant validity of the constructs for this research context.

As a whole, the measurement model analysis demonstrated adequate convergent validity and discriminant validity to bring this research to the next level: structural model analysis for testing the relationship among variables in the research model. The results of the reflective measurement model for this research is summarized in Table 4.

D. STRUCTURAL MODEL ANALYSIS

After the indicators of the constructs are confirmed to be reliable and valid, the next step is to perform structural model analysis. The structural model assesses the model in terms of predictive capabilities and relationships between the constructs. The PLS-SEM approach maximizes the explained variance of the endogenous construct; for this research, it is the intention to adopt EA. Following Hair *et al.* [156], [161]'s guideline and suggestion, six measures— significance of collinearity, path coefficient (β), coefficient of determination (R²), effect size (f²), blindfolding and predictive relevance (Q²), and effect size (q²)—can be adopted to assess

TABLE 4. Summary of reflective measurement model.

Constructs	T.	Converge	nt Validity	Internal Consiste	ncy Reliability	Disc	riminar	t Validity
	Items	OL > 0.70	AVE > 0.50	CR > 0.7	$\alpha > 0.7$	FL	CL	HTMT 0.90
COMM	COMM1	0.835	0.777	0.943	0.954	Yes	Yes	Yes
	COMM2	0.920					Yes	
	COMM3	0.927					Yes	
	COMM4	0.897					Yes	
	COMM5	0.866					Yes	
	COMM6	0.839					Yes	
CP	CP1	0.914	0.864	0.922	0.950	Yes	Yes	Yes
	CP2	0.943					Yes	
	CP3	0.932					Yes	
CPX	CPX1	0.871	0.804	0.923	0.942	Yes	Yes	Yes
	CPX2	0.869					Yes	
	CPX3	0.928					Yes	
	CPX4	0.916					Yes	
EB	EB1	0.960	0.926	0.980	0.984	Yes	Yes	Yes
	EB2	0.957					Yes	
	EB3	0.965					Yes	
	EB4	0.964					Yes	
	EB5	0.966					Yes	
EXT	EXT1	0.830	0.629	0.856	0.894	Yes	Yes	Yes
	EXT2	0.798					Yes	
	EXT3	0.702					Yes	
	EXT4	0.827					Yes	
	EXT5	0.802					Yes	
GVR	GVR1	0.873	0.834	0.960	0.968	Yes	Yes	Yes
	GVR2	0.879					Yes	
	GVR3	0.901					Yes	
	GVR4	0.947					Yes	
	GVR5	0.935					Yes	
100	GVR6	0.941	0.000			•••	Yes	
ICI	ICTI	0.814	0.680	0.767	0.864	Yes	Yes	Yes
	ICT2	0.866					Yes	
	IC13	0.791					Yes	
INT	INT1	0.975	0.943	0.970	0.980	Yes	Yes	Yes
	INT2	0.977					Yes	
	INT3	0.960					Yes	
MP	MP1	0.965	0.931	0.963	0.976	Yes	Yes	Yes
	MP2	0.973					Yes	
	MP3	0.956					Yes	
NP	NP1	0.943	0.856	0.916	0.947	Yes	Yes	Yes
	NP2	0.893					Yes	
	NP3	0.939					Yes	
OR	OR1	0.836	0.594	0.746	0.848	Yes	Yes	Yes
	OR2	0.865					Yes	
	OR3	0.844					Yes	
	OR4	0.465					No	
OS	CAT1	1.000	1.000	1.000	1.000	Yes	Yes	Yes
TMS	TMS1	0.900	0.816	0.925	0.946	Yes	Yes	Yes
	TMS2	0.884					Yes	
	TMS3	0.904					Yes	
	TMS4	0.924					Yes	

OL - outer loadings; FL = Fornell and Larcker; CL = cross-loading; CR = composite reliability

the structural model by using PLS-SEM. Next, importanceperformance map analysis (IPMA) was performed to identify the factors that have a relatively high influence on EA adoption by MPS organizations. IPMA is an advanced method and a useful tool but is less frequently used in research [161], [163]. An illustrative outcome of IPMA enables researchers to identify the critical constructs of attention and action. Finally, the summary of the hypotheses of this research context is presented.

1) COLLINEARITY ASSESSMENT

Collinearity assessment typically involves assessing the formative measurement model. This step is necessary in structural model assessment. According to Hair *et al.* [147], collinearity for the structural model estimates the path

VOLUME 8, 2020

coefficient based on ordinary least squares regression of each endogenous construct on its corresponding exogenous construct. A collinearity issue occurs when two exogenous constructs are causally related to each other. The criterion of collinearity is VIF, and the threshold value of VIF is below 5 (VIF < 5) [147]. Every construct involved in this research was examined using this measure. The critical level of collinearity (VIF > 5) may indicate the biasness in the path coefficient for multiple regression and does not reliably explain the endogenous construct or dependent variable in the research model. Although the VIF values of EB and GVR are near the critical levels of multicollinearity, multicollinearity is not a serious issue for a predictive model of dependent variables such as the intention to adopt EA [141], [164], [165] because multicollinearity does not affect the reliability of prediction. The value of VIF suggested by Sekaran and Bougie [141],

TABLE 5. VIF result.

Construct	INT
COMM	1.733
CP	2.414
CPX	1.178
EB	4.956
EXT	1.919
GVR	4.596
ICT	1.766
MP	2.373
NP	3.98
OR	2.308
SIZE	1.096
TMS	2.258

Yu [164], Ho [165] is lower than 10 (VIF < 10). Table 5 shows the result of the collinearity analysis of the model. All the inner VIF values of all constructs are lower than 10 and 5. Therefore, no multicollinearity issue exists for the research model.

2) PATH COEFFICIENT

Path coefficient (β) assesses the hypothesized relationships among the constructs in the research model. This research has a nonparametric distribution. Thus, both measures of t-value and path coefficients (β) were obtained using the bootstrapping procedure. The standardized values of the path coefficient are approximately within the range ± 1 . The path coefficient relies on its standard error by running the bootstrapping procedure. Consequently, the standard error enables the confidence interval, t-value, and p-value to be computed and exhibited for all structural path coefficients. The bootstrapping procedure of 500 samples, no sign changes, basic bootstrapping, bias-corrected and accelerated, one-tailed test, and significance level of 0.05 were selected to provide consistent values by using SmartPLS 3.0. The critical value of t-value is equal to or greater than 1.65 (t > 1.65), and the p-value is lower than 0.05 (p < 0.05) [147]. In this research, 12 direct hypotheses were developed between constructs, and the results of the path coefficients are presented in Table 6. Seven relationships have a t-value ≥ 1.65 and are hence significant at 0.05. The factors are clear communication ($\beta =$ -0.133, p < 0.01), coercive pressure ($\beta = 0.330$, p < 0.01), expected benefit ($\beta = 0.180$, p < 0.05), good governance $(\beta = 0.237, p < 0.05)$, mimetic pressure ($\beta = 0.130, p$ < 0.05), normative pressure ($\beta = 0.165$, p < 0.01), and organization size ($\beta = 0.07$, p < 0.05). Therefore, H1, H2, H4, H6, H8, H9, and H11 are supported.

Interestingly, complexity (CPX), external support (EXT), sufficient ICT infrastructure (ICT), organizational readiness (OR), and top management support (TMS) have insignificant relationships with EA adoption by MPS public sector organizations. Thus, H3 with $\beta = -0.018$, p < 0.05, H5 with $\beta = -0.039$, p < 0.05, H7 with $\beta = 0.004$, p < 0.05, H10



FIGURE 3. Hypothesis testing.

with $\beta = -0.010$, p < 0.05, and H12 with $\beta = 0.022$, p < 0.05 are not supported. Figure 3 shows the path coefficients with p-values on each relationship defined for this research model.

3) COEFFICIENT OF DETERMINATION (R2)

Another important measure for assessing the structural model is the coefficient of determination (\mathbb{R}^2) . This measure explains the model's predictive power and represents the combinations' effect of exogenous constructs on the endogenous construct [147]. The criterion of the coefficient of determination is R^2 because it represents the values of square correlations between exogenous constructs and a specific endogenous construct. Therefore, the R² defines a measure of in-sample predictive power and ranges between 0 and 1 [156]. Although scholars claimed that providing rules of thumb for an acceptable \mathbb{R}^2 value is difficult, the [159], [166] guidelines were adopted. The R^2 values of 0.75, 0.50, or 0.25 are defined as high, moderate, or weak, respectively. The high value of R² indicates a high predictive accuracy. The R² value of the structural model for this research is 0.665, which is above the 0.20 and 0.26 values suggested by [167]. The result indicates a substantial model for this research. Scholars also claimed that R² values should be high enough to achieve a minimum level of explanatory power, and a value greater than 0.10 is reasonably adequate to explain the variance of the endogenous construct [28], [168].

4) EFFECT SIZE F2

The next measure is the effect size of exogenous constructs toward the endogenous construct by using Cohen f^2 [167]. This measure is highly recommended and encouraged by scholars and reviewers [28], [156], [158], [159], [169]. A guideline for assessing the effect size (f^2) is that values of 0.35, 0.15, and 0.02 indicate large, medium and small effects [167] of the exogenous constructs, respectively.

TABLE 6. Hypothesis testing.

Η	Relationship	SB (_β)	SE	t-value	p-value	LL (0.05)	UL (0.95)	Supported
H1	COMM -> INT	-0.133	0.054	2.459	0.007	-0.213	-0.033	Yes
H2	CP -> INT	0.330	0.075	4.387	0.000	0.208	0.459	Yes
H3	CPX -> INT	-0.018	0.047	0.376	0.353	-0.095	0.057	No
H4	EB -> INT	0.180	0.090	2.002	0.023	0.032	0.327	Yes
Н5	EXT -> INT	-0.039	0.061	0.643	0.26	-0.139	0.054	No
H6	GVR -> INT	0.237	0.102	2.314	0.011	0.060	0.391	Yes
H7	ICT -> INT	0.004	0.049	0.089	0.465	-0.076	0.088	No
H8	MP -> INT	0.130	0.074	1.760	0.040	0.009	0.257	Yes
Н9	NP -> INT	0.165	0.079	2.076	0.019	0.041	0.298	Yes
H10	OR -> INT	-0.010	0.062	0.157	0.438	-0.119	0.085	No
H11	SIZE -> INT	0.07	0.035	1.968	0.025	0.009	0.127	Yes
H12	TMS -> INT	0.022	0.058	0.376	0.353	-0.066	0.120	No

Note: H = hypothesis, SB = standard beta, SE = standard error, ** p < 0.01, * p < 0.05

TABLE 7. Result of effect size (F²).

	INT	F^2
COMM	0.034	Small
СР	0.152	Medium
CPX	0.000	None
EB	0.022	Small
EXT	0.001	None
GVR	0.042	Small
ICT	0.000	None
MP	0.021	Small
NP	0.021	Small
OR	0.002	None
SIZE	0.000	None
TMS	0.000	None

TABLE 8. Result of predictive relevance (Q²) and effect size (Q²).

Construct	Q^2	Q^2 Included	Q^2	q^2	Effect
			Excluded		Change
COMM		0.575	0.565	0.024	Small
СР		0.575	0.527	0.114	Medium
CPX		0.575	0.576	-0.001	No effect
EB		0.575	0.570	0.013	No effect
EXT		0.575	0.571	0.010	No effect
GVR		0.575	0.562	0.031	Small
ICT		0.575	0.576	-0.001	No effect
INT	0.575	0.575			
	(High)				
MP		0.575	0.571	0.010	No effect
NP		0.575	0.570	0.013	No effect
OR		0.575	0.575	0.001	No effect
SIZE		0.575	0.571	0.010	No effect
TMS		0.575	0.581	-0.013	No effect

A value lower than 0.02 indicates no effect on the endogenous construct. The effect size's result for the exogenous constructs in this research is presented in Table 7. The exogenous construct CP has a medium size effect, while COMM, EB, GVR, MP, and NP have a small size effect toward the endogenous construct (INT). Exogenous constructs (CPX, EXT, ICT, OR, SIZE and TMS) have no effect size toward the endogenous construct (INT).

5) PREDICTIVE RELEVANCE (Q2)

In addition to R^2 values as a criterion of predictive accuracy power, the predictive relevance criterion should be examined for assessing the structural model [170], [171]. Unlike R^2 , Q^2 is also referred to as the out-sample predictive power due to its ability to predict data that are not used in the model. Therefore, the blindfolding procedure depending on the omission distance was applied to provide the Q² value of endogenous constructs that have a reflective measurement model [147]. The blindfolding algorithm is based on the cross-validated redundancy approach and was defined with omission distance = 7, path weighting scheme, 500 maximum iterations, and stop criterion = 7. The result of the predictive relevance of endogenous constructs in this research is presented in Table 8. The Q² value for endogenous construct (INT) is 0.575, which is greater than 0. Therefore, all the exogenous constructs have predictive relevance for the endogenous construct. Also, as a rule of thumb, Hair *et al.* [147] stated that the values of 0.35, 0.15, and 0.02 represent large, medium, and small predictive relevance of endogenous constructs, respectively.

6) EFFECT SIZE (Q2)

The relative impact of predictive relevance was assessed by using a similar approach of effect size for R^2 values. The q^2 effect size values were computed using the formula

$$q^{2} = (Q_{Included}^{2} - Q_{Excluded}^{2})/(1 - Q_{Included}^{2})$$
(1)

A relative measure of predictive relevance values of 0.35, 0.15, and 0.02 represents large, medium, and small predictive relevance, respectively, of exogenous constructs toward the endogenous construct for this research context [147]. As Table 8 shows, a medium q2 effect size is found for CP, while a small q2 effect size is observed for COMM and GVR. Other exogenous constructs have no effect size of predictive relevance toward the endogenous construct (INT).

7) IMPORTANCE-PERFORMANCE MAP ANALYSIS

The IPMA contrasts the total effects of the structural model, which is the endogenous construct (INT) with the average scores of exogenous constructs [154], [158], [170]. The total effects represent the importance of the exogenous construct in influencing the endogenous construct, while their average scores of exogenous constructs represent their performance. The exogenous constructs that have a strong total effect perform relatively well for the endogenous construct. However, low performance is observed with low average scores of exogenous constructs.

Table 9 provides the result of IPMA under study by using IPMA procedure in SmartPLS 3.0. The construct of EB (70.227) performs well in influencing the intention to adopt EA (INT). The result also indicates that a one-unit index value for the performance of EB increase results in an increased overall performance of target construct (INT) by the value of the total effect of EB (0.187). Although the exogenous construct of CP has a high importance in explaining the target construct because its total effect data is 0.313, the performance of CP is relatively low, thereby leaving substantial room for improvement. Consequently, the construct CP is the most relevant for managerial actions. On the other hand, the construct EB is the most relevant for managerial attention. The IPMA of intention to adopt EA (INT) reveals that the construct COMM also has high performance. However, it is not an important construct in the prediction of intention to adopt EA (INT). Therefore, managers or decision-makers should not put much focus and attention on COMM because it may result in possible overkill in public sector services in the case of EA.

The researchers plotted the index values and total effects scores in a priority map, as shown in Figure 4. the more important constructs are CP, GVR, EB, NP, and MP. Therefore, managers or decision-makers should focus on these constructs to enhance adoption. The interpretation of IPMA is consistent with [154], [156]. The construct OS has little influence on EA adoption due to its low importance and performance. The results of IPMA allow the identification of factors with a relatively high importance and relatively low performance. These are key improvement areas that can

TABLE 9. Result of the IPMA.

Constructs	Total Effect (Important)	Index Value (Performances)
COMM	-0.148	70.051
CP	0.313	59.242
EB	0.187	70.227
GVR	0.26	68.311
MP	0.135	62.188
NP	0.165	63.916
OS	0.054	45.392



FIGURE 4. IPMA (priority map) for EA adoption among MPS organizations.

be addressed by management activities afterwards. However, insignificant constructs CPX, EXT, ICT, OR, and TMS are excluded from further IPMA because the basic requirements of PLS-SEM assessment were not met [163].

VI. DISCUSSION

This research examined the factors that influence EA adoption by the MPS and identified these factors by designing and modeling the organizational view to achieve desirable IT solutions. EA adoption by the public sector is dependent on clear communication, coercive pressure, expected benefit, good governance, mimetic pressure, normative pressure, and organizational size. Surprisingly, top management support, ICT infrastructure, EA complexity, organizational readiness, and external support have a nonsignificant influence on the intention to adopt EA by public sector organizations. The following discussion is based on the perspectives of technology, organization, and environment.

A. TECHNOLOGICAL CONTEXT

Unexpectedly, sufficient ICT infrastructure does not associate with EA adoption by MPS, as indicated by the value of the path coefficient ($\beta = 0.004$) and probability (p > 0.05). This finding is inconsistent with past studies on the adoption of e-government [58], SaaS [62], IoT [64], and ERP [108], where this factor was a significant factor influencing IS/IT adoption in different contexts that use the TOE framework. This finding is not surprising because they have not adopted EA yet, so they might not be aware of the technical requirement of EA. Furthermore, the responses were collected among five different MPS organization types, from the federal level to the local authority. For the state-level organizations, their ICT infrastructure is coordinated by the federal or central agency in the state, such as the office of the state secretary. Therefore, they cannot decide on the sufficiency of ICT infrastructure in their organization.

Complexity has an insignificantly negative influence on EA adoption by MPS organizations, as indicated by the value of path coefficient ($\beta = -0.018$) and probability (p > 0.05). However, the finding of the research is also inconsistent with previous research where many organizations are concerned about the adoption of new IT innovations, particularly EA. One possible reason complexity is not considered a barrier to EA implementation in MPS is that many agencies might not be aware and have adequate knowledge on the subject. They may think and assume that the EA approach is similar to the ISP and are not aware that EA uses a dedicated tool to manage the organization's architecture. The EA tool designs and models the complexity of the organization from the business, information, application, and technology perspectives, thereby requiring skilled architects. Designing the organization view involves a detailed process and critical thinking. Another possible reason is the availability of third-party support such as EA vendors, who fully appreciate the new trend and technology needs of the public sector. Furthermore, the initiator of EA, such as MAMPU, has been promoting EA since 2014. This condition could be another reason EA complexity negatively influences EA adoption in MPS.

B. ORGANIZATIONAL CONTEXT

Remarkably, top management support has no significant influence on EA adoption, as indicated by the value of path coefficient ($\beta = 0.022$) and probability (p > 0.05). This result is inconsistent with most previous studies, which found that top management support has a positive relationship with IT/IS adoption such as technology adoption [108], cloud computing [172], social CRM [173], and IoT [64]. However, past studies also found that top management is a barrier or has a negative influence on IS/IT adoption, such as RFID in the retail industry [174], RFID in manufacturing [57], and mobile reservation systems [175]. The reason for this insignificant result is that most of the respondents are non-adopters and are thus indistinguishable in terms of the factor of top management support. The factor of top management support that leads to the unreadiness of MPS decision-makers to adopt EA to their service operation is their level of understanding about the EA concept as a whole.

Most MPS organizations use the ISP as their guide and plan to implement desirable IT solutions. For them, EA is ISP, which they believe they already have. The real understanding of the EA concept is a problem for them. Consequent to this notion, they cannot understand the reason behind the adoption decision; they are not ready to spend or allocate resources and are already satisfied with only the ISP. Another possible reason is that EA is still in its infancy, as reported by [17]. Furthermore, due to the lack of a common standard of EA adoption and implementation in the public sector, top management would rather adopt the wait-and-see approach by looking at other agencies that have adopted EA, developed EA, and successfully implemented and validated EA projects. MPS organization types are generally under the exclusive purview of the state and federal governments; for example, a state statutory body is under the purview of the state government. A local authority is under the purview of the state government and the Ministry of Urban Wellbeing, Housing, and Local Government. These types of purview are indistinguishable in terms of the factor of top management support. Thus, top management support would be an insignificant factor in EA adoption by MPS organizations.

The relationship between organizational readiness and EA adoption by MPS organizations is insignificant ($\beta = -0.010$, t = 0.157, p = 0.438). This empirical result is inconsistent with findings from past studies, which established the significant role of organization readiness, e.g., halal warehouse by Ngah et al. [54], business intelligence systems [176], and IoT [64]. Motivated MPS organizations that want to adopt a new IS/IT initiative must have the technical ability, organizational support, and experienced people and skill before new IS/IT initiative adoption is possible [177], [178]. In this case, even the predictive relevance to the intention to adopt EA by MPS organizations is high ($Q^2 = 0.575$, mean = from 7-point Likert scale). EA is a detailed process, which implies the need for skilled and experienced people in EA and organizational business to design and model the organizational viewpoints and desirable IT solutions for the organization. Organizational readiness means that the organization is expected to have the technological and managerial resources and skills to adopt EA more intensively and pervasively. The study also shows that financial situation is one of the factors that lead to the unreadiness of the adoption of a new IS/IT initiative. Skilled and experienced people are a time-sensitive and monetary investment. The majority of the respondents claimed that the top management did not provide any financial support for adopting the EA and training their employees to have a better understanding of and competency in EA. This idea is supported by the result of top management support (H3) factor in this research.

The findings of this research show that clear communication is a significantly positive influence on the intention to adopt EA in the MPS context ($\beta = 0.133$, t = 2.459, p = 0.007). As suggested by EA scholars, EA has to be established among organizational members and communicated clearly to gain a mutual understanding of the adoption of EA [179], [180]. To be successful, EA should be accepted at all other work levels, including engineering, technical, and worker levels. This success requires clear EA communication, which includes a process of informing and obtaining facilitation from stakeholders about all EA-related issues [181]. EA should not just take place in IT or offices of the CIO, but it also requires strong recognition from the highest level of the organization [182]. Hence, EA stakeholders' participation is necessary, which means stakeholders should be involved in decision-making [183]. Ultimately, clear communication is viewed as the driver and voice of any initiative in the organization, thereby ensuring that EA will be more appreciated and accepted in the organization [184]. Therefore, clear communication is an important factor for EA adoption in the MPS context.

Remarkably, the normative pressure factor from institutional theory has a connection with EA adoption, as indicated by the value of $\beta = 0.165$, t = 2.076, p = 0.019 derived from the path coefficient analysis. The finding is similar to that of adoption studies in ERP [108], EA in Vietnam [67], IT governance [185], and social media [186]. This result may be explained by the fact that neglecting normative pressure may be operationally counterproductive in the organization. These findings imply that normative pressure is relatively more important than top management support and organization readiness in influencing the intention to adopt EA. A possible explanation for these results is that the extent to which an organization engages in cloud computing services and resources is still substantively reliant on the demands of professional associates, organizational culture, and the extent to which the government promotes the use of EA. Although MPS organizations are composed of different organization types that have varying organizational cultures and environments, they agree that normative pressure is a significant factor that influences EA adoption.

The expected benefit is revealed to have a significant positive influence on EA adoption by MPS organizations $(\beta = 0.180, t = 2.002, p = 0.023)$. This result is consistent with past studies on the adoption of halal warehouse by Ngah et al. [54], business intelligence systems [176], hospital information system (HIS) [49], IoT [64], and cloud computing [187]. In this research, MPS organizations are interested in adopting EA if they are expected to benefit from it. A possible explanation for this situation might be the role of the organization types of the public sector. Willingness to adopt EA depends on how EA can clearly manage the complexity of the organization, eliminate the segmentation of process and application, and deliver better services to citizens. In this regard, when the expected benefits are high and convincing, EA adoption rates increase. The belief is that using EA guarantees a holistic view of organizational strategies and desirable IT solutions. Consequently, EA should increase the efficiency of service delivery and citizens' confidence in the public sector. The initiator of EA, such as MAMPU, should aggressively promote EA by explaining its benefits to the stakeholders and the decision-making team of organizations. Such promotion could be conducted by organizing a talk session and awareness program regarding the importance and benefits of adopting EA, especially during the Public Sector CIO Convex because the target participants are the CIOs of public sector organizations and the majority of them are non-adopters of EA. Government agencies should help convey the message that EA obtains more benefits if they

98864

change from an unstructured view of the organization to the EA approach. As a result, it will help the decision-makers decide on a strategic organizational goal [188].

The relationship between good governance and EA adoption is significant, as indicated by $\beta = 0.237$, t = 2.314, and p = 0.011 derived from the path coefficient analysis. The empirical result also accords with earlier literature inferences, which showed that the role of good governance is a determinant of EA studies such as those on service-oriented EA (SOEA) [189], the deployment of EA in the Namibian government [84], EA in IS controls [190], and transformation readiness of EA [77]. Good governance in EA affects EA adoption by MPS organizations as a result of the complexity of the organizational structure of the public sector. The complexity of the organization consists of multiple viewpoints, including business, information, technology, and system. Different architects design and model these viewpoints accordingly to obtain a holistic and strategic view of the organizational goal. To accomplish this, the identification of stakeholders, roles and responsibilities, systematic procedures, and ongoing commitment across organizational members and stakeholders related to EA are necessary. Therefore, most decision-makers in public sector organizations are willing to adopt EA when clear accountability, roles, and responsibility for decision-making are available. In addition, EA implementation has to be measured for its impact on the organization. Only then can EA be realized and cultivated by establishing good governance.

The findings of this research show that organizational size is an important factor for EA adoption in the MPS context, as indicated by the value of $\beta = 0.07$, t = 1.968, and p = 0.025 derived from the path coefficient analysis. The empirical result is consistent with the recent previous studies in EA [13], HIS adoption [49], e-government assimilation [58], and mobile hotel reservation system adoption [175]. This result may be explained by the fact that large organizations benefit more because they have more manpower and support to invest in and adopt EA compared with smaller organizations. Another possible explanation for this situation is that, due to the need of the government for digital transformation in the public sector and the greater demand from citizens, a sizeable amount of manpower should accelerate the process of decision-making of EA adoption. Organization size plays an important role as a facilitator because having sufficient resources enables organizations to accommodate any consequences of steps taken by the decision-makers or stakeholders. As a result, the confidence level of the decision-makers at larger organizations is higher, making it easier to measure the implication of EA and enhance organizational innovations. Furthermore, larger organizations exhibit value creation by developing their attractiveness and motivation among the decision-makers and then influence organizational members to accept innovation regardless of the organization's size.

The IT department size represents the technical resources of the organization to adopt EA because EA needs skilled architects to design and model the viewpoints of application, system, and information. On the basis of the demographic profile of the number of IT employees in the organization, the majority of the respondents (51.8%) have fewer than 10 IT personnel. From the percentage of the IT department size of this research, the researcher can infer that having a small number of IT employees may increase innovation resistance; hence, they refuse to accept EA. Therefore, MPS organizations seem to agree that IT department size is important in the adoption of EA because a big department size corresponds to the wide technological base of the organization for initiating and implementing IS innovations [191].

C. ENVIRONMENTAL CONTEXT

The external support factor has an insignificant impact on EA adoption by MPS organizations ($\beta = -0.039$, t = 0.643, p = 0.26). This outcome is contrary to that of MacLennan and Van Belle [82], Ahmadi, et al. [49], Rababah [192], and Sophonthummapharn [55], who found that external support was significant to the SOA, HIS, CRM, and techno-relationship innovation adoption, respectively. This contradictory result may be due to the effect of external support on the adoption of cloud computing, especially from vendors, which may be different for each organization. Not all organizations received support from the same vendors for the same solutions provided. To the researcher's knowledge, Malaysia has only one dominant EA vendor. Hence, the cost of skill and competency development, such as training, certification, and consultation, is high. This scenario indirectly influences the demotivation among decision-makers to adopt EA. An EA tool is a must in EA for designing and modeling the architecture. It is available in the market as both open-source and proprietary software. For example, the ArchiMate language can perform modeling, but it requires paid training and a certified EA trainer. These relationships may partly be explained by the inadequate training from The National Institute of Public Administration (INTAN) and the MAMPU. INTAN is the training arm of the Public Service Department, Malaysia, which offers a fundamental course on EA but not the technical part of EA to model the architecture. These agencies may not be able to accommodate demand, especially for certification and coaching, from MPS organizations. Another possible explanation is that the inherent characteristics of the agencies that have sufficient internal expertise makes them independent of external support for EA activities and resource implementation. The external support becomes insignificant in this research due to the lack of awareness of external support available in the industry. MPS organizations might not be aware of what EA requirements have to be in place and of the industry's overall practices. Hence, the dependency of EA adoption on the external support of the MPS organizations increases with awareness.

As shown in the previous section, the relationship between coercive pressure and EA adoption by MPS organizations is significant ($\beta = 0.330$, t = 4.387, p = 0.00). This result corroborates the findings of a great deal of previous works on IS/IT adoption studies, such as OGD by.Wang and

Lo [59], HIS by Nilashi et al. [193], supply chain management system by Liu et al. [194], and e-commerce technology by Kurnia et al. [195]. Specifically, this factor reflects that of Duong [67], who also found that coercive pressure is an important pressure in institutionalizing EA in the Vietnam public sector. Hence, this argument supports previous research into this area, which links coercive pressure and EA adoption by MPS organizations. A possible explanation for this finding might be the pressure perceived by virtue of the competitive conditions, requirements, and incentives from the federal or local government and industry associations in MPS [49]. The requirements are raised by political, regulatory, and constituents' influences, including citizens and the federal and central governments. As an important knowledge source for its agencies, powerful organizations such as federal, state, and central agencies can pressure agencies under their exclusive purview by raising requirements. In this research, the IPMA results proved that the factor of coercive pressure is the one that most influenced EA adoption by MPS organizations. This result is likely to be related to the decision-making of MPS organizations, which considers written and formal directions from superior parties such as the central government as desirable, so that EA can be seen as important and beneficial to the public sector in general. Therefore, a possible suggestion is that coercive pressure is a key improvement area that can be addressed by management activities afterwards. Further studies that take this factor into account will need to be undertaken.

Coercive pressure is viewed as the promotion, direction, and support from MAMPU as the main agency for IT administration and policymakers in MPS. This research found that promotion, direction, and support from MAMPU influence other agencies in the MPS to explore the potential benefits of EA. This result may be explained by the asymmetry of power in the MPS structure, in which an organization's dependent agencies, such as a local authority or a state statutory body, under the exclusive purview of the state government may comply with such a requirement or policy to secure their competitive condition. In the case of EA adoption, when the federal and state governments require the agencies under their purview to adopt EA, the top management in the agencies may coercively push themselves to adopt EA as well. In this scenario, with a policy being put in place to support EA, resource support in terms of budget, training, consultation and other incentives are attainable for the future. Going against the requirements of the powerful organization may jeopardize the dependent organization's survival due to its dependence on the powerful organization. Therefore, the dependent organization tends to comply with the powerful organization under its exclusive purviews' requirements and be motivated to adopt EA.

The findings also revealed that mimetic pressure is important for EA adoption by MPS organizations, as indicated by the value of $\beta = 0.130$, t = 1.76, and p = 0.04 derived from the path coefficient analysis. This result reflects that of Duong [67], who also found that mimetic pressure

TABLE 10.	Prior studies	on ea	adoption	and the	toe	framework.
-----------	---------------	-------	----------	---------	-----	------------

						EA	adc	ptic	on s	tudi	es										Pr	evic	us :	stud	ies	tha	t us	ed T	ΌĒ	r r			
Factor	[68]	[35]	[6]	[30]	[82]	[103]	[19]	[190]	[31]	[84]	[32]	**[83]	**[77]	[189]	[99]	* [54]	[57]	[58]	[59]	[60]	[108]	[64]	[113]	*[200]	[55]	* [192]	*[49]	[59]	**[121]	*[111]	**[185]	**[186]	[201] [175]
ICT infrastructure																																	
EA complexity																																	V V
Expected benefit																																	
Top management support																																	$\sqrt{}$
Organizational readiness																																	V
Clear communication																																	V
Normative pressure																																V	
Organization size																																	$\sqrt{}$
Good governance																																	
External IS support																																	
Mimetic pressure																																	
Coercive pressure																																	

*Malaysian context, **MPS

influences the adoption of EA in the Vietnam public sector. In addition, this factor corroborates the findings of a great deal of previous work on supply chain management system by Liu et al. [194], open platform adoption by Shim et al. [60], EDI use by Hart and Saunders [124], HIS by Ahmadi et al. [49], and SOEA [189]. These types of research are instances of complex innovation of mimetic pressure to conform to other organizations caused by the extent of EA adoption or by the perceived success of EA adoption by other organizations in the same industry [49]. In this research, such perception reflects the organization's perception of the environment and its reasonable status. Several possible explanations are given for this result. The non-adopter of EA will benchmark and credit the exemplary organization's success to its strategic alternatives and imitate this successful organization by embracing the same practices. Hence, the particular organization can reduce its risks and costs. This logic can be proffered to the context of the organization's EA adoption decision. The organization may learn how other government agencies benefit from EA and perceive mimetic pressures to imitate these successful agencies [194].

Furthermore, this example of a successful agency impacts the level of certainty among MPS decision-makers in adopting EA and may reduce the cost of investigating the value of EA. In this research context of the public sector, cost is a motivation to adopt EA. MPS organizations may accept the mimetic pressure and be inclined to adopt EA. Another possible reason is that mimetic pressures play a role in this research because MPS organizations perceive EA as complex to understand and use. This argument is supported by the insignificant relationship between EA complexity and EA adoption (H2). Public sector organizations differ according to the type of organization structure and are thus likely to be unaware of the benefits of EA. Therefore, they may be prone to mimetic pressure. Such reasons may explain why this research found support for the positive effect of mimetic pressures on organizational intention to adopt EA.

VII. RESEARCH IMPLICATIONS

This research has three implications: theoretical, contextual, and practical. First, the development of a new EA adoption

model contributes to a new theoretical finding in the field of EA that focuses on the technology, organization, and environment contexts based on the TOE framework as the underpinning theory and the integration of elements of institutional theory. This research is the first to examine the technological, organizational, and environmental factors that influence public sector adoption of EA. This research identifies seven key factors of public sector adoption of EA: clear communication, coercive pressure, expected benefit, good governance, mimetic pressure, normative pressure, and organizational size. In addition, the quantitative approach of model validation using PLS-SEM has made a significant contribution as empirical research in the EA research domain, because most EA studies use qualitative approaches such as case studies and interviews [21], [26]. This achievement is considered a significant contribution to the body of knowledge because quantitative approach is perceived as a structured way to generalize the population as a whole by examining the relationship between variables [132], [141]. Therefore, the findings of this research could be generalized by other public sectors in other countries that have a similar government structural composition as Malaysia.

Second, the context of this research is the MPS. The use of IT and IS in the public sector context has received continuous interest in past research, but most of the studies were conducted in developed countries such as Denmark, the United States, Finland, and Germany. Surprisingly, literature examining the adoption of IT/IS in developing countries is lacking. This research discussed the gap in knowledge by examining the determinants of EA adoption in a developing country such as Malaysia, particularly in the public sector context. Clear communication and good governance tend to be specific factors among the seven factors that influence the public sector's adoption of EA, particularly in the Malaysian context.

Third, the research results allow for useful practical contributions. The participation of EA and public sector practitioners as experts in this analysis has made the research findings sufficiently valid for identifying real-world phenomena. The findings of this research will support EA initiators in the MPS and other developing countries in designing new EA



TABLE 11. Measurement items of this research.

Variables	Code	Measurement items	Source
	EXT3	Agencies outside your organization provide training on EA.	
	EXT4	Technology vendors outside your organization actively market EA by providing incentives for adoption.	
	EXT5	Technology vendors outside your organization promote EA by offering free training–workshop	
Complexity	CPX1	Your organization believes that EA is complex to use.	[57]
1 5	CPX2	Your organization believes that EA is a complex process.	
	CPX3	The skills required to use EA are too complex for your employees.	[204]
Environte 11 au eCta	CPX4	Integrating EA in your current work practices will be very difficult.	[101
Expected benefits	EBI FB2	EA enables your organization to accomplish tasks more quickly. EA improves the quality of your organization's work	[121, 2051
	EB2 EB3	EA makes it easier to do your organization's job.	205]
	EB4	EA improves your organization's service delivery performance.	
	EB5	EA enhances your organization's effectiveness in service delivery.	
Organization size		Number of employees:	[191]
		< 100	
		100-399	
		400–699	
		700-999	
		> 1000	
Top management	TMS1	The top management team is highly interested in using EA.	[204]
support	TMS2	The top management team will allocate adequate resources for the adoption of the EA.	
	TMS5 TMS4	The top management team has a vision as a leader to promote EA to the organization.	
Organizational	OR1	Your organization gave its staff a formal explanation regarding halal supply chain services.	[206],
readiness	OR2	Your organization has knowledgeable staff to adopt EA.	[55]
	OR3	Your organization has the financial resources to adopt EA.	
Clear	OR4 COMM1	Budget was the important factor that the organization had to deal with before adopting EA activities.	[207]
communication	COMMIN	In your organization, everyone has a chance to express their opinion.	[207]
communication	COMM2	Your organization listens to each organizational member.	
	COMM3	In your organization, employees feel free to make positive and negative comments.	
	COMM4	In your organization, even though organizational members do not have total agreement,	
	COMM5	organizational members do reach a kind of consensus that everyone in the organization accepts.	
Good governance	GVR1	The achievement of an EA comes from the persistent responsibility taken	
Good governance	GVR2	The initiative of implementing an EA can definitely identify decision-making accountability.	[112]
	GVR3	You have thoroughly analyzed the outcome of adopting EA, which may cause some changes in the	
	61 m (organization, suppliers, business partners, and customers.	
	GVR4	Y ou have followed a systematic procedure for dealing with changes caused by the implementation of	
	GVR5	You have clearly defined the business cases for every initiative of EA.	
	GVR6	You have clearly defined a measure to evaluate the impact of adopting EA.	
Normative pressure	NP1	The organization believes that you should use EA.	[208]
	NP2	EA is the norm in your industry.	
Mimetic pressure	NP3 MP1	Using EA is beneficial to the organization.	[10/]
Winnette pressure	MP2	Your main competitors who have adopted EA are favorably perceived by others in the same industry.	[194]
	MP3	Your main competitors who have adopted EA are more competitive.	
Coercive pressure	CP1	The government requires your organization to use EA.	[109]
	CP2	The industry association requires your organization to use EA.	
Organization type	CP3	Competitive conditions require your organization to use EA.	[58]
organization type		Federal	[50]
		Federal statutory body	
		Local authority	
		State	
		State statutory body	
		State statutory body	

initiatives. EA initiators can now identify the key factors that need to be considered for EA implementation to ensure that

this initiative is widely accepted in the future by the public sector. Policy makers can use the research findings as inputs

in the creation of EA checklists to allow horizontal sharing and incorporation of knowledge across multiple public sector organizations. This research also assists the top management in public sector organizations in strategizing EA adoption by recognizing the important factors of EA adoption in their organization.

VIII. RESEARCH LIMITATIONS AND FUTURE WORK

The context of this research is restricted to the MPS, consisting of federal, federal statutory body, state, state statutory body, and local organizations. Although this research covered the population of MPS organizations, the generalizability of these results is subject to certain limitations. For instance, the results showed that the complexity of EA, ICT infrastructure, top management support, and external support have nonsignificant influences on EA adoption by MPS organizations. Given that these results are inconsistent with previous studies on public sector IT/IS adoption, the researcher suggests that any future work should include the moderation effect in the relationship between these variables in EA adoption because a moderator variable is usually added when the relationship between a variable and a criterion variable is unexpectedly weak or inconsistent [147].

IX. CONCLUSION

The benefits of EA include eliminating system duplication and silos [72], aligning IT and business in the organization for strategic planning and investment [98], [196], [197], reducing complexity in the IT infrastructure [198], [199], and improving business agility and dynamic change [66]. However, despite the perceived benefits of EA, public sector adoption remains sluggish, as demonstrated by scarce EA adoption studies in the existing literature. The goal of this research was to develop and validate a theoretical model that analyzed the influence of TOE factors on the adoption of EA by the MPS. An unanticipated finding was the weak influence of complexity of EA, ICT infrastructure, top management support, and external support on EA adoption by MPS organizations. Together, the research findings add to the knowledge base in the EA area and may contribute to better insights for EA initiators in their EA implementation plan and strategic planning to support EA implementation among MPS organizations.

APPENDIX A

See Table 10.

APPENDIX B

See Table 11.

ACKNOWLEDGEMENT

This research is supported by The Energy University (UNITEN).

REFERENCES

- [1] Enterprise Architecture as a Digital Transformation Facilitator: 2018 Results From EA Survey, McKinsey, Gurgaon, India, 2018.
- [2] The Evolution of Enterprise Architecture, Gartner, Stamford, CT, USA, 2019.
- [3] Y. Masuda and M. Viswanathan, "Direction of digital IT and enterprise architecture," in *Enterprise Architecture for Global Companies in a Digital IT Era*. Springer, 2019, pp. 17–59.
- [4] M. Zhang, H. Chen, and A. Luo, "A systematic review of business-IT alignment research with enterprise architecture," *IEEE Access*, vol. 6, pp. 18933–18944, 2018.
- [5] O. Moscoso-Zea, J. Castro, J. Paredes-Gualtor, and S. Luján-Mora, "A hybrid infrastructure of enterprise architecture and business intelligence & analytics for knowledge management in education," *IEEE Access*, vol. 7, pp. 38778–38788, 2019.
- [6] C. Economics. (Feb. 2017). Enterprise Architecture Not Widely Embraced. Accessed: Mar. 26, 2018. [Online]. Available: https://www. computereconomics.com/article.cfm?id=2330
- [7] United Nations E-Government Survey 2010 Leveraging E-Government at a Time Of Financial and Economic Crisis, U. Nations, New York, NY, USA, 2010.
- [8] P. Saha, Enterprise Architecture for Connected E-Government: Practices and Innovations: Practices and Innovations. Hershey, PA, USA: IGI Global, 2012.
- [9] V. Seppänen, "From problems to critical success factors of enterprise architecture adoption," *Jyväskylä Stud. Comput.*, no. 201, 2014.
- [10] D. D. Dang and S. Pekkola, "Problems of enterprise architecture adoption in the public sector: Root causes and some solutions," in *Information Technology Governance in Public Organizations* (Integrated Series in Information Systems). Cham, Switzerland: Springer, 2017, ch. 8, pp. 177–198.
- [11] N. Syynimaa, "Method and practical guidelines for overcoming enterprise architecture adoption challenges," in *Enterprise Information Systems* (Lecture Notes in Business Information Processing). Cham, Switzerland: Springer, 2017, ch. 22, pp. 488–514.
- [12] S. Kar and R. Thakurta, "Planning for digital transformation: Implications for institutional enterprise architecture," *Planning*, no. 6, p. 26, 2018.
- [13] J. J. Korhonen and M. Halen, "Enterprise architecture for digital transformation," presented at the IEEE 19th Conf. Bus. Inform. (CBI), Jul. 2017.
- [14] N. A. Abu Bakar, S. Yaacob, S. S. Hussein, A. Nordin, and H. Sallehuddin, "Dynamic metamodel approach for government enterprise architecture model management," *Procedia Comput. Sci.*, vol. 161, pp. 894–902, Jan. 2019.
- [15] The Malaysian Public Sector Ict Strategic Plan 2016–2020, MAMPU, Berlin, Germany, 2016.
- [16] S. S. Hussein, M. N. Mahrin, and N. Maarop, "Sustainability through innovations of enterprise architecture (EA) in public sector's management: Issues & challenges," *J. Southeast Asian Res.*, vol. 2017, no. 2017, pp. 1–13, 2017, doi: 10.5171/2017.722027.
- [17] Tahap Kesediaan Enterprise Architecture Sektor Awam Siri #2, MAMPU, Berlin, Germany, 2016.
- [18] D. D. Dang and S. Pekkola, "Root causes of enterprise architecture problems in the public sector," in *Proc. PACIS*, Jun. 2016, p. 287.
- [19] N. Syynimaa, Enterprise Architecture Adoption Method for Higher Education Institutions. Reading, U.K.: Univ. Reading, 2015.
- [20] D. Dang, T. Vartiainen, and S. Pekkola, "Patterns of enterprise architecture adoption in the public sector: A resource-based perspective," in *Proc.* 27th Eur. Conf. Inf. Syst., 2019, pp. 1–17.
- [21] N. A. Ahmad, S. Mohd. Drus, and N. A. Abu Bakar, "Enterprise architecture adoption issues and challenges: A systematic literature review," *Indonesian J. Electr. Eng. Comput. Sci.*, vol. 15, no. 1, p. 399, Jul. 2019, doi: 10.11591/ijeecs.v15.i1.pp399-408.
- [22] M. Lange, J. Mendling, and J. Recker, "An empirical analysis of the factors and measures of enterprise architecture management success," *Eur. J. Inf. Syst.*, vol. 25, no. 5, pp. 411–431, Sep. 2016, doi: 10.1057/ejis.2014.39.
- [23] N. A. A. Bakar and H. Selamat, "Investigating enterprise architecture implementation in public sector organisation: A case study of ministry of health Malaysia," presented at the 3rd Int. Conf. Comput. Inf. Sci. (ICCOINS), Aug. 2016.

- [24] B. D. Rouhani, R. B. Ahmad, F. Nikpay, and R. Mohamaddoust, "Critical success factor model for enterprise architecture implementation," *Malaysian J. Comput. Sci.*, vol. 32, no. 2, pp. 133–148, Apr. 2019, doi: 10.22452/mjcs.vol32no2.4.
- [25] M. Ylinen and S. Pekkola, "Enterprise architecture as a scapegoat for difficulties in public sector organizational transformation," in *Proc. 39th Int. Conf. Inf. Syst.*, San Francisco, CA, USA, 2018, pp. 1–13.
- [26] N. A. Ahmad, S. M. Drus, N. A. A. Bakar, and M. M. Othman, "A systematic review of enterprise architecture adoption models," *Int. J. Eng. Technol.*, vol. 7, no. 4, pp. 369–374, 2018, doi: 10.14419/ijet.v7i4.35. 22763.
- [27] N. A. A. Bakar, N. Kama, and H. Selamat, "Enterprise architecture development and implementation in public sector: The Malaysian perspective," J. Theor. Appl. Inf. Technol., vol. 88, no. 1, 2016.
- [28] N. Urbach and F. Ahlemann, "Structural equation modeling in information systems research using partial least squares," J. Inf. Technol. Theory Appl., vol. 11, no. 2, pp. 5–40, 2010.
- [29] M. Bernaert, G. Poels, M. Snoeck, and M. De Backer, "CHOOSE: Towards a metamodel for enterprise architecture in small and medium-sized enterprises," *Inf. Syst. Frontiers*, vol. 18, no. 4, pp. 781–818, Aug. 2016, doi: 10.1007/s10796-015-9559-0.
- [30] M. Bernaert, G. Poels, M. Snoeck, and M. De Backer, "Enterprise architecture for small and medium-sized enterprises: A starting point for bringing ea to SMEs, based on adoption models," in *Information Systems* for Small and Medium-sized Enterprises. Berlin, Germany: Springer, 2014.
- [31] S. Gilliland, P. Kotze, and A. van der Merwe, "Work level related human factors for enterprise architecture as organisational strategy," presented at the Int. Conf. Enterprise Syst. (ES), Oct. 2015.
- [32] F. Behrouz and M. Fathollah, "A systematic approach to enterprise architecture using axiomatic design," *Procedia CIRP*, vol. 53, pp. 158–165, 2016, doi: 10.1016/j.procir.2016.07.012.
- [33] Systems and Software Engineering-Architecture Description, I. I. I., Mumbai, India, 2011, 2011.
- [34] M. Lankhorst, Enterprise Architecture at Work: Modelling, Communication and Analysis. Springer, 2009.
- [35] K. Hiekkanen, J. J. Korhonen, J. Collin, E. Patricio, M. Helenius, and J. Mykkanen, "Architects' perceptions on EA use—An empirical study," in *Proc. IEEE 15th Conf. Bus. Informat.*, Jul. 2013, pp. 292–297.
- [36] N. Syynimaa, "The quest for underpinning theory of enterprise architecture–General systems theory," in *Proc. 19th Int. Conf. Enterprise Inf. Syst.*, 2017, pp. 1–9.
- [37] I. Ajzen, "The theory of planned behavior," Org. Behav. Hum. Decis. Process., vol. 50, no. 2, pp. 179–211, 1991.
- [38] F. D. Davis, R. P. Bagozzi, and P. R. Warshaw, "User acceptance of computer technology: A comparison of two theoretical models," *Manage. Sci.*, vol. 35, no. 8, pp. 982–1003, Aug. 1989.
- [39] V. Venkatesh and F. D. Davis, "A theoretical extension of the technology acceptance model: Four longitudinal field studies," *Manage. Sci.*, vol. 46, no. 2, pp. 186–204, Feb. 2000.
- [40] V. Venkatesh, M. G. Morris, G. B. Davis, and F. D. Davis, "User acceptance of information technology: Toward a unified view," *MIS Quart.*, vol. 27, no. 3, pp. 425–478, Sep. 2003.
- [41] S. Taylor and P. A. Todd, "Understanding information technology usage: A test of competing models," *Inf. Syst. Res.*, vol. 6, no. 2, pp. 144–176, Jun. 1995.
- [42] R. Depietro, E. Wiarda, and M. Fleischer, "The context for change: Organization, technology and environment," *Processes Technol. Innov.*, vol. 199, no. 0, pp. 151–175, 1990.
- [43] M. R. Everett, *Diffusion of Innovations*, vol. 12. New York, NY, USA: The Free Press of Glencoe, 1995.
- [44] W. R. Scott, "Institutional theory: Contributing to a theoretical research program," *Great Minds Manage.*, *Process Theory Develop.*, vol. 37, pp. 460–484, Jan. 2005.
- [45] M. Wade and J. Hulland, "The resource-based view and information systems research: Review, extension, and suggestions for future research," *MIS Quart.*, vol. 28, no. 1, pp. 107–142, 2004.
- [46] J. Baker, "The technology-organization-environment framework," in Information Systems Theory, Springer, 2012, pp. 231–245.

- [47] J. A. Miles, Management and Organization Theory (The Jossey-Bass Business and Management Reader Series). Hoboken, NJ, USA: Wiley, 2012.
- [48] P. DiMaggio and W. W. Powell, "The iron cage revisited: Collective rationality and institutional isomorphism in organizational fields," *Amer. Sociol. Rev.*, vol. 48, no. 2, pp. 147–160, 1983.
- [49] H. Ahmadi, M. Nilashi, L. Shahmoradi, and O. Ibrahim, "Hospital information system adoption: Expert perspectives on an adoption framework for malaysian public hospitals," *Comput. Hum. Behav.*, vol. 67, pp. 161–189, Feb. 2017, doi: 10.1016/j.chb.2016.10.023.
- [50] W. H. DeLone and E. R. McLean, "The DeLone and McLean model of information systems success: A ten-year update," J. Manage. Inf. Syst., vol. 19, no. 4, pp. 9–30, 2003.
- [51] Y. K. Dwivedi, M. R. Wade, and S. L. Schneberger, *Information Systems Theory Explaining and Predicting Our Digital Society*, vol. 1. New York, NY, USA: Springer, 2012. [Online]. Available: http://www.springer.com/series/6157
- [52] A. Tarhini, N. A. G. Arachchilage, R. Masa'deh, and M. S. Abbasi, "A critical review of theories and models of technology adoption and acceptance in information system research," *Int. J. Technol. Diffusion*, vol. 6, no. 4, pp. 58–77, Oct. 2015.
- [53] F. Damanpour and M. Schneider, "Phases of the adoption of innovation in organizations: Effects of environment, organization and top Managers1," *Brit. J. Manage.*, vol. 17, no. 3, pp. 215–236, Sep. 2006.
- [54] A. H. Ngah, Y. Zainuddin, and R. Thurasamy, "Applying the TOE framework in the halal warehouse adoption study," *J. Islamic Accounting Bus. Res.*, vol. 8, no. 2, pp. 161–181, Apr. 2017, doi: 10.1108/jiabr-04-2014-0014.
- [55] K. Sophonthummapharn, "The adoption of techno-relationship innovations," *Marketing Intell. Planning*, vol. 27, no. 3, pp. 380–412, May 2009, doi: 10.1108/02634500910955254.
- [56] T. S. H. Teo, S. Lin, and K.-H. Lai, "Adopters and non-adopters of e-procurement in singapore: An empirical study," *Omega*, vol. 37, no. 5, pp. 972–987, Oct. 2009, doi: 10.1016/j.omega.2008.11.001.
- [57] Y.-M. Wang, Y.-S. Wang, and Y.-F. Yang, "Understanding the determinants of RFID adoption in the manufacturing industry," *Technol. Forecasting Social Change*, vol. 77, no. 5, pp. 803–815, Jun. 2010, doi: 10.1016/j.techfore.2010.03.006.
- [58] B. Pudjianto, H. Zo, A. P. Ciganek, and J. J. Rho, "Determinants of e-government assimilation in indonesia: An empirical investigation using a toe framework," *Asia Pacific J. Inf. Syst.*, vol. 21, no. 1, pp. 49–80, 2011.
- [59] H.-J. Wang and J. Lo, "Adoption of open government data among government agencies," *Government Inf. Quart.*, vol. 33, no. 1, pp. 80–88, Jan. 2016.
- [60] S. Shim, B. Lee, and S. L. Kim, "Rival precedence and open platform adoption: An empirical analysis," *Int. J. Inf. Manage.*, vol. 38, no. 1, pp. 217–231, Feb. 2018, doi: 10.1016/j.ijinfomgt.2017.10.001.
- [61] A. E. Youssef and A. M. Mostafa, "Critical decision-making on cloud computing adoption in organizations based on augmented force field analysis," *IEEE Access*, vol. 7, pp. 167229–167239, 2019.
- [62] Z. Yang, J. Sun, Y. Zhang, and Y. Wang, "Understanding SaaS adoption from the perspective of organizational users: A tripod readiness model," *Comput. Hum. Behav.*, vol. 45, pp. 254–264, Apr. 2015, doi: 10.1016/j.chb.2014.12.022.
- [63] M. Mukred, Z. M. Yusof, F. M. Alotaibi, U. A. Mokhtar, and F. Fauzi, "The key factors in adopting an electronic records management system (ERMS) in the educational sector: A UTAUT-based framework," *IEEE Access*, vol. 7, pp. 35963–35980, 2019.
- [64] C.-W. Hsu and C.-C. Yeh, "Understanding the factors affecting the adoption of the Internet of Things," *Technol. Anal. Strategic Manage.*, vol. 29, no. 9, pp. 1089–1102, Oct. 2017, doi: 10.1080/09537325.2016.1269160.
- [65] A. J. W. Chen, M. Boudreau, and R. T. Watson, "Information systems and ecological sustainability," *J. Syst. Inf. Technol.*, vol. 10, no. 3, pp. 186–201, Nov. 2008.
- [66] D. Dang and S. Pekkola, "Enterprise architecture and organizational reform: A project debrief," in *Proc. 21st Pacific Asian Conf. Inf. Syst. PACIS*, 2017, p. 71.
- [67] P. S. D. Duong, "Institutionalising enterprise architecture in the public sector in Vietnam," in *Proc. Assoc. Inf. Syst. AIS Electron. Library* (AISeL), 2016, pp. 1–16, Paper Summer 6-15-2016. [Online]. Available: http://aisel.aisnet.org/ecis2016_rp/139

- [68] K. H. Madsen, "Institutional patterns of enterprise architecture adoption in government," *Transforming Government, People, Process Policy*, vol. 1, no. 4, pp. 333–349, Dec. 2007, doi: 10.1108/17506160710839169.
- [69] B. Kuo and G. Dick, "The greening of organisational IT: What makes a difference?" Australas. J. Inf. Syst., vol. 16, no. 2, pp. 81–92, 2010.
- [70] N. Syynimaa, "Modeling the dynamics of enterprise architecture adoption process," in *Proc. Int. Conf. Enterprise Inf. Syst.*, Springer, 2015, pp. 577–594.
- [71] H. Ahmad, A. Ahmad, and N. Hazlina Nik Abdullah Hafaz, "Conceptual knowledge relationship model of enterprise architecture and top management roles," 2016, arXiv:1606.01449. [Online]. Available: http://arxiv.org/abs/1606.01449
- [72] F. Gamiet, Enterprise Architecture (EA) as a Governance Tool to Reduce Application Duplication Study of a Duplication: A Case South African Provincial Government. Cape Town, South Africa: Univ. Western Cape, 2012.
- [73] A. Østergaard Jensen, "Government enterprise architecture adoption," M.S. thesis, Copenhagen Bus. School, Copenhagen, Denmark, 2010. [Online]. Available: http://studenttheses.cbs.dk/bitstream/handle/ 10417/1736/anders_oestergaard_jensen.pdf
- [74] R. A. Razak and Z. Md Dahalin, "An exploratory study of enterprise architecture practices in Malaysia," *Commun. IBIMA*, vol. 3, pp. 133–137, Jun. 2008.
- [75] R. A. Razak, Z. M. Dahalin, H. Ibrahim, N. I. Yusop, and M. K. Kasiran, "Investigation on the importance of enterprise architecture in addressing business issues," in *Proc. Int. Conf. Res. Innov. Inf. Syst.*, Nov. 2011, pp. 1–4.
- [76] N. A. A. Bakar, S. Harihodin, and N. Kama, "An assessment model for government enterprise architecture establishment phase," *Adv. Sci. Lett.*, vol. 20, no. 10, pp. 1987–1991, Oct. 2014.
- [77] S. S. Hussein, M. N. R. Mahrin, and N. Maarop, "Preliminary study of Malaysian public sector (MPS) transformation readiness through enterprise architecture (EA) establishment," in *Proc. 21st Pacific Asia Conf. Inf. Syst.*, Langkawi, Malaysia, 2017, pp. 1–11.
- [78] B. Harian, "EA pacu inovasi digital perkhidmatan kerajaan," *Berita Harian*, Kuala Lumpur, Malaysia, 2019.
- [79] Digital government in Malaysia, MAMPU, Berlin, Germany, 2015.
- [80] K. Zhu and K. L. Kraemer, "Post-adoption variations in usage and value of E-business by organizations: Cross-country evidence from the retail industry," *Inf. Syst. Res.*, vol. 16, no. 1, pp. 61–84, Mar. 2005.
- [81] C.-H. Yeh, G.-G. Lee, and J.-C. Pai, "Using a technology-organizationenvironment framework to investigate the factors influencing e-business information technology capabilities," *Inf. Develop.*, vol. 31, no. 5, pp. 435–450, Nov. 2015, doi: 10.1177/0266666913516027.
- [82] E. MacLennan and J.-P. Van Belle, "Factors affecting the organizational adoption of service-oriented architecture (SOA)," *Inf. Syst. e-Bus. Manage.*, vol. 12, no. 1, pp. 71–100, Feb. 2014, doi: 10.1007/s10257-012-0212-x.
- [83] N. A. Abu Bakar, S. Harihodin, and N. Kama, "Service-oriented enterprise architecture (SOEA) adoption and maturity measurement model: A systematic literature review," *Int. J. Comput., Inf. Sci. Eng.*, vol. 7, pp. 15–27, Nov. 2013.
- [84] I. Shaanika and T. Iyamu, "Deployment of enterprise architecture in the namibian government: The use of activity theory to examine the influencing factors," *Electron. J. Inf. Syst. Developing Countries*, vol. 71, no. 1, pp. 1–21, Nov. 2015.
- [85] Z. I. S. S. Hussein and M. N. Mahrin, "EA innovations in managing public sectors: Issues & challenges," J. Southeast Asian Res., vol. 2017, no. 722027, pp. 3498–3505, 2016, doi: 10.5171/2017.722027.
- [86] V. Venkatesh and H. Bala, "Adoption and impacts of interorganizational business process standards: Role of partnering synergy," *Inf. Syst. Res.*, vol. 23, no. 4, pp. 1131–1157, Dec. 2012.
- [87] M. H. Meyer and K. F. Curley, "The impact of knowledge and technology complexity on information systems development," *Expert Syst. Appl.*, vol. 8, no. 1, pp. 111–134, Jan. 1995.
- [88] F. Parveen and A. Sulaiman, "Technology complexity, personal innovativeness and intention to use wireless Internet using mobile devices in Malaysia," *Int. Rev. Bus. Res. Papers*, vol. 4, no. 5, pp. 1–10, 2008.
- [89] R. Ayyagari, V. Grover, and R. Purvis, "Technostress: Technological antecedents and implications," *MIS Quart.*, vol. 35, no. 4, pp. 831–858, Dec. 2011.

- [90] B. Wejnert, "Integrating models of diffusion of innovations: A conceptual framework," *Annu. Rev. Sociology*, vol. 28, no. 1, pp. 297–326, Aug. 2002.
- [91] R. Agarwal, "Individual acceptance of information technologies," in Framing the Domains of IT Management: Projecting the Future Through the Past. Cincinnati, OH, USA: Pinnaflex Education Resources, 2000, pp. 85–104.
- [92] R. Sharma and P. Yetton, "The contingent effects of management support and task interdependence on successful information systems implementation," *MIS Quart.*, vol. 27, no. 4, pp. 533–556, Dec. 2003.
- [93] A. Balaid, M. Z. A. Rozan, and S. N. Abdullah, "Conceptual model for examining knowledge maps adoption in software development organizations," *Asian Social Sci.*, vol. 10, no. 15, p. 118, Jul. 2014.
- [94] E. T. G. Wang, G. Klein, and J. J. Jiang, "ERP misfit: Country of origin and organizational factors," *J. Manage. Inf. Syst.*, vol. 23, no. 1, pp. 263–292, Jun. 2006.
- [95] L. G. Tornatzky and K. J. Klein, "Innovation characteristics and innovation adoption-implementation: A meta-analysis of findings," *IEEE Trans. Eng. Manag.*, vols. EM–29, no. 1, pp. 28–45, 1982.
- [96] G. Zaltman, R. Duncan, and J. Holbek, *Innovations and Organizations*. Hoboken, NJ, USA: Wiley, 1973.
- [97] H. Sallehudin, R. C. Razak, and M. Ismail, "Factors influencing cloud computing adoption in the public sector: An empirical analysis," *J. Entrepreneurship Bus.*, vol. 3, no. 1, pp. 30–45, Jun. 2015.
- [98] K. Hjort-Madsen, Architecting Government: Understanding Enterprise Architecture Adoption in the Public Sector. Copenhagen, Denmark: IT-Universitetet i København, 2009.
- [99] F. Nikpay, R. Ahmad, and C. Yin Kia, "A hybrid method for evaluating enterprise architecture implementation," *Eval. Program Planning*, vol. 60, pp. 1–16, Feb. 2017, doi: 10.1016/j.evalprogplan.2016.09.001.
- [100] D. H. Olsen and K. Trelsgård, "Enterprise architecture adoption challenges: An exploratory case study of the norwegian higher education sector," *Procedia Comput. Sci.*, vol. 100, pp. 804–811, 2016, doi: 10.1016/j.procs.2016.09.228.
- [101] C. L. Iacovou, I. Benbasat, and A. S. Dexter, "Electronic data interchange and small organizations: Adoption and impact of technology," *MIS Quart.*, vol. 19, no. 4, pp. 465–485, Dec. 1995.
- [102] R. Tulenheimo, "Challenges of implementing new technologies in the world of BIM—Case study from construction engineering industry in finland," *Procedia Econ. Finance*, vol. 21, pp. 469–477, Jan. 2015.
- [103] B. T. Hazen, L. Kung, C. G. Cegielski, and L. A. Jones-Farmer, "Performance expectancy and use of enterprise architecture: Training as an intervention," *J. Enterprise Inf. Manage.*, vol. 27, no. 2, pp. 180–196, Feb. 2014, doi: 10.1108/jeim-08-2012-0042.
- [104] R. Sharma and P. Yetton, "The contingent effects of training, technical complexity, and task interdependence on successful information systems implementation," *MIS Quart.*, vol. 31, no. 2, pp. 219–238, Jun. 2007.
- [105] J. Lemmetti and S. Pekkola, "Understanding enterprise architecture: Perceptions by the finnish public sector," in *Proc. Int. Conf. Electron. Government.* Springer, 2012, pp. 162–173.
- [106] Y. Gong and M. Janssen, "The value of and myths about enterprise architecture," *Int. J. Inf. Manage.*, vol. 46, pp. 1–9, Jun. 2019.
- [107] S. Aier, "The role of organizational culture for grounding, management, guidance and effectiveness of enterprise architecture principles," *Inf. Syst. e-Bus. Manage.*, vol. 12, no. 1, pp. 43–70, Feb. 2014, doi: 10.1007/s10257-012-0206-8.
- [108] H. O. Awa, O. U. Ojiabo, and L. E. Orokor, "Integrated technologyorganization-environment (T-O-E) taxonomies for technology adoption," *J. Enterprise Inf. Manage.*, vol. 30, no. 6, pp. 893–921, Oct. 2017, doi: 10.1108/jeim-03-2016-0079.
- [109] H. Liang, N. Saraf, Q. Hu, and Y. Xue, "Assimilation of enterprise systems: The effect of institutional pressures and the mediating role of top management," *MIS Quart.*, vol. 31, no. 1, pp. 59–87, Mar. 2007.
- [110] S. A. S. Weiss and R. Winter, "Institutionalization and the effectiveness of enterprise architecture management," presented at the 34th Int. Conf. Inf. Syst., Milan, Italy, 2013.
- [111] F. T. S. Chan and A. Y.-L. Chong, "Determinants of mobile supply chain management system diffusion: A structural equation analysis of manufacturing firms," *Int. J. Prod. Res.*, vol. 51, no. 4, pp. 1196–1213, Feb. 2013.

- [112] W.-H. Hung, L.-M. Chang, C.-P. Lin, and C.-H. Hsiao, "E-readiness of website acceptance and implementation in SMEs," *Comput. Hum. Behav.*, vol. 40, pp. 44–55, Nov. 2014, doi: 10.1016/j.chb.2014.07.046.
- [113] S. Krishnan, T. S. Teo, and J. Lim, "E-participation and e-government maturity: A global perspective," in *Proc. Int. Work. Conf. Transf. Diffusion IT.* Springer, 2013, pp. 420–435.
- [114] H. Mohammad Kazem, L. Christine, and A. Frederik, Beyond EA Frameworks: Towards an Understanding of The Adoption of Enterprise Architecture Management. Atlanta, GA, USA: Association for Information Systems AIS Electronic Library (AISeL), 2012. [Online]. Available: http://aisel.aisnet.org/ecis2012
- [115] P. Meso, P. Datta, and V. Mbarika, "Moderating information and communication technologies' influences on socioeconomic development with good governance: A study of the developing countries," *J. Amer. Soc. Inf. Sci. Technol.*, vol. 57, no. 2, pp. 186–197, Jan. 2006.
- [116] M. C. Lemos and A. Agrawal, "Environmental governance and political science," in *Governance for the environment: New Perspectives* (Annu. Rev. Environ. Resour), no. 31. 2009, pp. 297–325, doi: 10.1146/annurev.energy.31.042605.135621.
- [117] P. Saha, "Architecting the connected government: Practices and innovations in singapore," in *Proc. 3rd Int. Conf. Theory Pract. Electron. Governance ICEGOV*, 2009, pp. 11–17.
- [118] S. Krishnan and T. S. H. Teo, "Moderating effects of governance on information infrastructure and e-government development," *J. Amer. Soc. Inf. Sci. Technol.*, vol. 63, no. 10, pp. 1929–1946, Oct. 2012.
- [119] S. Krishnan, "Electronic government maturity: Antecedents and consequences from a global perspective," Ph.D. dissertation, Inf. Syst., Nat. Univ. Singapore, Singapore, Tech. Rep., 2014.
- [120] J. Y. L. Thong, "An integrated model of information systems adoption in small businesses," J. Manage. Inf. Syst., vol. 15, no. 4, pp. 187–214, Mar. 1999.
- [121] H. Sallehudin, R. C. Razak, and M. Ismail, "Determinants and impact of cloud computing implementation in the public sector," J. Adv. Inf. Technol., vol. 7, pp. 245–251, Nov. 2016.
- [122] J. Benamati and A. L. Lederer, "Rapid information technology change, coping mechanisms, and the emerging technologies group," *J. Manage. Inf. Syst.*, vol. 17, no. 4, pp. 183–202, 2001.
- [123] T. M. Somers and K. Nelson, "The impact of critical success factors across the stages of enterprise resource planning implementations," in *Proc. 34th Annu. Hawaii Int. Conf. Syst. Sci.*, Jan. 2001, p. 10.
- [124] P. J. Hart and C. S. Saunders, "Emerging electronic partnerships: Antecedents and dimensions of EDI use from the supplier's perspective," *J. Manage. Inf. Syst.*, vol. 14, no. 4, pp. 87–111, Mar. 1998.
- [125] Q. Hu and J. Quan, "The institutionalization of IT budgeting: Empirical evidence from the financial sector," *Inf. Resour. Manage. J.*, vol. 19, no. 1, pp. 84–97, Jan. 2006.
- [126] N. A. A. Bakar and H. Selamat, "Investigating enterprise architecture implementation in public sector organisation: A case study of ministry of health malaysia," in *Proc. 3rd Int. Conf. Comput. Inf. Sci. (ICCOINS)*, Aug. 2016, pp. 1–6.
- [127] P. Tingling and M. Parent, "Mimetic isomorphism and technology evaluation: Does imitation transcend judgment?" J. Assoc. Inf. Syst., vol. 3, no. 1, pp. 113–143, 2002.
- [128] E. Abrahamson and L. Rosenkopf, "Institutional and competitive bandwagons: Using mathematical modeling as a tool to explore innovation diffusion," *Acad. Manage. Rev.*, vol. 18, no. 3, pp. 487–517, Jul. 1993.
- [129] L. G. Zucker, "Institutional theories of organization," Annu. Rev. Sociology, vol. 13, no. 1, pp. 443–464, Aug. 1987.
- [130] M. Tina Dacin, J. Goodstein, and W. Richard Scott, "Institutional theory and institutional change: Introduction to the special research forum," *Acad. Manage. J.*, vol. 45, no. 1, pp. 45–56, Feb. 2002.
- [131] S. MacKenzie, P. Podsakoff, and N. Podsakoff, "Construct measurement and validation procedures in mis and behavioral research: Integrating new and existing techniques," *MIS Quart.*, vol. 35, no. 2, pp. 293–334, 2011.
- [132] J. W. Creswell, Educational Research: Planning, Conducting, and Evaluating Quantitative, 4th ed. Upper Saddle River, NJ, USA: Prentice-Hall, 2014.
- [133] M. J. Sanchez-Franco and J. L. Roldán, "Expressive aesthetics to ease perceived community support: Exploring personal innovativeness and routinised behaviour as moderators in tuenti," *Comput. Hum. Behav.*, vol. 26, no. 6, pp. 1445–1457, Nov. 2010, doi: 10.1016/j.chb.2010.04. 023.

- [134] L. J. Cronbach, "A research worker's treasure chest," *Multivariate Behav. Res.*, vol. 19, nos. 2–3, pp. 223–240, Apr. 1984.
- [135] I. Brace, Questionnaire Design: How to Plan, Structure and Write Survey Material for Effective Market Research. London, U.K.: Kogan Page Publishers, 2018.
- [136] P. Chwelos, I. Benbasat, and A. S. Dexter, "Research report: Empirical test of an EDI adoption model," *Inf. Syst. Res.*, vol. 12, no. 3, pp. 304–321, Sep. 2001.
- [137] M. L. Xu and S. O. Leung, "Effects of varying numbers of likert scale points on factor structure of the rosenberg self-esteem scale," *Asian J. Social Psychol.*, vol. 21, no. 3, pp. 119–128, Sep. 2018.
- [138] G. A. Miller, "The magical number seven, plus or minus two: Some limits on our capacity for processing information," *Psychol. Rev.*, vol. 63, no. 2, p. 81, 1955.
- [139] R. Hussein, N. S. A. Karim, N. Mohamed, and A. R. Ahlan, "The influence of organizational factors on information systems success in E-Government agencies in malaysia," *Electron. J. Inf. Syst. Developing Countries*, vol. 29, no. 1, pp. 1–17, Mar. 2007.
- [140] N. Podsakoff, "Common method biases in behavioral research: A critical review of the literature and recommended remedies," J. Appl. Psychol., vol. 885, no. 879, p. 10.1037, 2003.
- [141] U. Sekaran and R. Bougie, *Research Methods for Business: A Skill Building Approach*. Hoboken, NJ, USA: Wiley, 2016.
- [142] M.-C. Boudreau, D. Gefen, and D. W. Straub, "Validation in information systems research: A state-of-the-art assessment," *MIS Quart.*, vol. 25, no. 1, pp. 1–16, 2001.
- [143] A. Field, Discovering Statistics Using IBM SPSS Statistics. Thousand Oaks, CA, USA: SAGE, 2013.
- [144] D. George and P. Mallery, SPSS for Windows Step by Step: A Simple Guide and Reference 11.0 Update, 4th ed. Boston, MA, USA: Allyn & Bacon, 2003.
- [145] P. Kline, *The Handbook of Psychological Testing*, 2nd ed. London, U.K.: Routledge, 1999.
- [146] J. A. Gliem and R. R. Gliem, "Calculating, interpreting, and reporting Cronbach's alpha reliability coefficient for likert-type scales," in *Proc. Midwest Res.-Pract. Conf. Adult, Continuing, Community Educ.*, 2003, pp. 82–88.
- [147] J. F. Hair, Jr, G. T. M. Hult, C. Ringle, and M. Sarstedt, A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM). Thousand Oaks, CA, USA: SAGE Publications, 2016.
- [148] Y. Baruch and B. C. Holtom, "Survey response rate levels and trends in organizational research," *Human Relations*, vol. 61, no. 8, pp. 1139–1160, Aug. 2008.
- [149] D. Muijs, Doing Quantitative Research in Education With SPSS. Thousand Oaks, CA, USA: SAGE, 2010.
- [150] R. P. Bagozzi, Y. Yi, and L. W. Phillips, "Assessing construct validity in organizational research," *Administ. Sci. Quart.*, vol. 36, no. 3, pp. 421–458, Sep. 1991.
- [151] N. Kock, "Common method bias in PLS-SEM: A full collinearity assessment approach," *Int. J. e-Collaboration*, vol. 11, no. 4, pp. 1–10, Oct. 2015.
- [152] N. K. Malhotra, S. S. Kim, and A. Patil, "Common method variance in IS research: A comparison of alternative approaches and a reanalysis of past research," *Manage. Sci.*, vol. 52, no. 12, pp. 1865–1883, Dec. 2006.
- [153] B. G. Tabachnick, L. S. Fidell, and J. B. Ullman, Using Multivariate Statistics. Boston, MA, USA: Pearson, 2007.
- [154] J. Hair, C. L. Hollingsworth, A. B. Randolph, and A. Y. L. Chong, "An updated and expanded assessment of PLS-SEM in information systems research," *Ind. Manage. Data Syst.*, vol. 117, no. 3, pp. 442–458, Apr. 2017.
- [155] T. Ramayah, J. Yeap, N. H. Ahmad, H. A. Halim, and S. A. Rahman, "Testing a confirmatory model of facebook usage in smartpls using consistent pls," *Int. J. Bus. Innov.*, vol. 3, no. 2, pp. 1–14, 2017.
- [156] J. F. Hair, J. J. Risher, M. Sarstedt, and C. M. Ringle, "When to use and how to report the results of PLS-SEM," *Eur. Bus. Rev.*, vol. 31, no. 1, pp. 2–24, Jan. 2019.
- [157] J. T. Roscoe, Fundamental Research Statistics for the Behavioral Sciences, 2nd ed. New York, NY, USA: Holt, Rinehart and Winston, Inc., 1975.
- [158] C. M. Ringle, M. Sarstedt, and D. Straub, "A critical look at the use of PLS-SEM in MIS quarterly," *MIS Quart.*, vol. 36, no. 1, pp. iii–xiv, 2012.

- [159] J. Henseler, C. M. Ringle, and R. R. Sinkovics, "The use of partial least squares path modeling in international marketing," in *New Challenges to International Marketing*. Bingley, U.K.: Emerald Group Publishing Ltd., 2009, pp. 277–319.
- [160] C. Fornell and D. Larcker, "A second generation of multivariate analysis: Classification of methods and implications for marketing research, review of marketing," *J. Houston, Amer. Marketing Assoc., Chicago*, no. 414, pp. 1–66, Jan. 1985.
- [161] J. F. Hair, M. Sarstedt, and C. M. Ringle, "Rethinking some of the rethinking of partial least squares," *Eur. J. Marketing*, vol. 53, no. 4, pp. 566–584, Apr. 2019.
- [162] A. H. Gold, A. Malhotra, and A. H. Segars, "Knowledge management: An organizational capabilities perspective," *J. Manage. Inf. Syst.*, vol. 18, no. 1, pp. 185–214, May 2001.
- [163] C. M. Ringle and M. Sarstedt, "Gain more insight from your PLS-SEM results: The importance-performance map analysis," *Ind. Manage. Data Syst.*, vol. 116, no. 9, pp. 1865–1886, Oct. 2016.
- [164] C. H. Yu, "An overview of remedial tools for collinearity in sas," in *Proc. Western Users SAS Softw. Conf.*, vol. 1, 2000, pp. 196–201.
- [165] R. Ho, Handbook of Univariate and Multivariate Data Analysis With IBM SPSS. Boca Raton, FL, USA: CRC Press, 2013.
- [166] J. F. Hair, C. M. Ringle, and M. Sarstedt, "PLS-SEM: Indeed a silver bullet," J. Marketing Theory Pract., vol. 19, no. 2, pp. 139–152, Apr. 2011.
- [167] E. Cohen, "Authenticity and commoditization in tourism," Ann. Tourism Res., vol. 15, no. 3, pp. 371–386, Jan. 1988.
- [168] R. F. Falk and N. B. Miller, A Primer for Soft Modeling. Akron, OH, USA: Univ. Akron Press, 1992.
- [169] D. Straub and D. Gefen, "Validation guidelines for IS positivist research," *Commun. Assoc. Inf. Syst.*, vol. 13, no. 1, p. 24, 2004.
- [170] S. Geisser, "A predictive approach to the random effect model," *Biometrika*, vol. 61, no. 1, pp. 101–107, 1974.
- [171] M. Stone, "Cross-validatory choice and assessment of statistical predictions," J. Roy. Stat. Soc., Ser. B, Methodol., vol. 36, no. 2, pp. 111–133, 1974.
- [172] T. Oliveira, M. Thomas, and M. Espadanal, "Assessing the determinants of cloud computing adoption: An analysis of the manufacturing and services sectors," *Inf. Manage.*, vol. 51, no. 5, pp. 497–510, Jul. 2014, doi: 10.1016/j.im.2014.03.006.
- [173] A. Ahani, N. Z. A. Rahim, and M. Nilashi, "Forecasting social CRM adoption in SMEs: A combined SEM-neural network method," *Comput. Hum. Behav.*, vol. 75, pp. 560–578, Oct. 2017, doi: 10.1016/j. chb.2017.05.032.
- [174] M. Bhattacharya and S. F. Wamba, "A conceptual framework of RFID adoption in retail using toe framework," in *Technology Adoption and Social Issues: Concepts, Methodologies, Tools, and Applications.* Hershey, PA, USA: IGI Global, 2018, pp. 69–102.
- [175] Y.-S. Wang, H.-T. Li, C.-R. Li, and D.-Z. Zhang, "Factors affecting hotels' adoption of mobile reservation systems: A technologyorganization-environment framework," *Tourism Manage.*, vol. 53, pp. 163–172, Apr. 2016, doi: 10.1016/j.tourman.2015.09.021.
- [176] B. Puklavec, T. Oliveira, and A. Popovi, "Understanding the determinants of business intelligence system adoption stages," *Ind. Manage. Data Syst.*, vol. 118, no. 1, pp. 236–261, Feb. 2018, doi: 10.1108/imds-05-2017-0170.
- [177] P. Chwelos, I. Benbasat, and A. S. Dexter, "Empirical test of an EDI adoption model," *Inf. Syst. Res.*, vol. 12, no. 3, pp. 304–321, 2001.
- [178] H.-F. Lin, "Examining the factors influencing knowledge management system adoption and continuance intention," *Knowl. Manage. Res. Pract.*, vol. 11, no. 4, pp. 389–404, Nov. 2013.
- [179] N. A. A. Bakar, S. Harihodin, and N. Kama, "Influence factors in government enterprise architecture establishment process: A preliminary findings," in *Proc. 23rd Int.-Bus.-Inf.-Manage.-Assoc. Conf. Visio, Sustain. Growth, Econ. Develop., Global Competitiveness*, vols. 1–5, K. Soliman, Ed. Norristown, PA, USA: INT Business Information Management Assoc-Ibima, Jun. 2013, pp. 576–583.
- [180] S. Aier and J. Schelp, "A reassessment of enterprise architecture implementation," in *Proc. Service-Oriented Comput. ICSOC/ServiceWave Workshops*, Springer, 2009, pp. 35–47.
- [181] B. van der Raadt, M. Bonnet, S. Schouten, and H. van Vliet, "The relation between EA effectiveness and stakeholder satisfaction," J. Syst. Softw., vol. 83, no. 10, pp. 1954–1969, Oct. 2010.

- [182] C. Schmidt and P. Buxmann, "Outcomes and success factors of enterprise IT architecture management: Empirical insight from the international financial services industry," *Eur. J. Inf. Syst.*, vol. 20, no. 2, pp. 168–185, Mar. 2011.
- [183] M. Hauder, C. Schulz, S. Roth, and F. Matthes, "Organizational factors influencing enterprise architecture management challenges," in *Proc.* 21st Eur. Conf. Inf. Syst. (ECIS), Utrecht, The Netherland, no. 175, 2013, pp. 1–12.
- [184] T. Iyamu, "Implementation of the enterprise architecture through the zachman framework," J. Syst. Inf. Technol., vol. 20, no. 1, pp. 2–18, Mar. 2018, doi: 10.1108/jsit-06-2017-0047.
- [185] M. F. I. Othman, Barriers to the Adoption of Formal it Governance Practice: A Malaysian Case. Brisbane, Australia: Queensland Univ. Technology, 2016.
- [186] M. H. B. Sharif, "Social media adoption and impact in australian local government," Ph.D. dissertation, Bus. School, Univ. Adelaide, Adelaide, SA, Australia, 2015.
- [187] C. Low, Y. Chen, and M. Wu, "Understanding the determinants of cloud computing adoption," *Ind. Manage. Data Syst.*, vol. 111, no. 7, pp. 1006–1023, Aug. 2011.
- [188] W. F. Boh and D. Yellin, "Using enterprise architecture standards in managing information technology," J. Manage. Inf. Syst., vol. 23, no. 3, pp. 163–207, Dec. 2006, doi: 10.2753/mis0742-1222230307.
- [189] A. Alwadain, E. Fielt, A. Korthaus, and M. Rosemann, "Empirical insights into the development of a service-oriented enterprise architecture," *Data Knowl. Eng.*, vol. 105, pp. 39–52, Sep. 2016, doi: 10.1016/j.datak.2015.09.004.
- [190] W. A. Cram, M. K. Brohman, and R. B. Gallupe, "Addressing the control challenges of the enterprise architecture process," *J. Inf. Syst.*, vol. 29, no. 2, pp. 161–182, Aug. 2015, doi: 10.2308/isys-50913.
- [191] H. H. Teo, K. K. Wei, and I. Benbasat, "Predicting intention to adopt interorganizational linkages: An institutional perspective," *MIS Quart.*, pp. 19–49, 2003.
- [192] K. A. M. Rababah, "Framework for adoption of customer relationship management system in hospital," Universiti Utara Malaysia, 2012.
- [193] M. Nilashi, H. Ahmadi, A. Ahani, R. Ravangard, and O. B. Ibrahim, "Determining the importance of hospital information system adoption factors using fuzzy analytic network process (ANP)," *Technol. Forecasting Social Change*, vol. 111, pp. 244–264, Oct. 2016.
- [194] H. Liu, W. Ke, K. K. Wei, J. Gu, and H. Chen, "The role of institutional pressures and organizational culture in the firm's intention to adopt Internet-enabled supply chain management systems," *J. Operations Manage.*, vol. 28, no. 5, pp. 372–384, Sep. 2010, doi: 10.1016/ j.jom.2009.11.010.
- [195] S. Kurnia, J. Choudrie, R. M. Mahbubur, and B. Alzougool, "E-commerce technology adoption: A malaysian grocery SME retail sector study," *J. Bus. Res.*, vol. 68, no. 9, pp. 1906–1918, Sep. 2015.
- [196] J. Du Lee and Y. I. Kwon, "A study on strategy planning and outcome of ea in Korea," in *Proc. 15th Int. Conf. Adv. Commun. Technol. (ICACT)*, Jan. 2013, pp. 873–879.
- [197] S. Gregor, D. Hart, and N. Martin, "Enterprise architectures: Enablers of business strategy and IS/IT alignment in government," *Inf. Technol. People*, vol. 20, no. 2, pp. 96–120, Jun. 2007, doi: 10.1108/ 09593840710758031.
- [198] M. Lange and J. Mendling, "An experts' perspective on enterprise architecture goals, framework adoption and benefit assessment," presented at the IEEE 15th Int. Enterprise Distrib. Object Comput. Conf. Workshops, Aug. 2011.
- [199] P. Närman, H. Holm, D. Höök, N. Honeth, and P. Johnson, "Using enterprise architecture and technology adoption models to predict application usage," J. Syst. Softw., vol. 85, no. 8, pp. 1953–1967, 2012, doi: 10.1016/j.jss.2012.02.035.
- [200] J. Yan Xin, T. Ramayah, P. Soto-Acosta, S. Popa, and T. Ai Ping, "Analyzing the use of Web 2.0 for brand awareness and competitive advantage: An empirical study in the malaysian hospitability industry," *Inf. Syst. Manage.*, vol. 31, no. 2, pp. 96–103, Apr. 2014.
- [201] P. R. Palos-Sanchez, F. J. Arenas-Marquez, and M. Aguayo-Camacho, "Cloud computing (SaaS) adoption as a strategic technology: Results of an empirical study," *Mobile Inf. Syst.*, vol. 2017, pp. 1–20, Jun. 2017.
- [202] C.-J. Chen, M. B. Gregoire, S. Arendt, and M. C. Shelley, "College and University dining services administrators' intention to adopt sustainable practices: Results from us institutions," *Int. J. Sustainability Higher Edu.*, vol. 12, no. 2, pp. 145–162, 2011.

- [203] G. Premkumar and K. Ramamurthy, "The role of interorganizational and organizational factors on the decision mode for adoption of interorganizational systems," *Decis. Sci.*, vol. 26, no. 3, pp. 303–336, May 1995.
- [204] G. Premkumar and M. Roberts, "Adoption of new information technologies in rural small businesses," *Omega*, vol. 27, no. 4, pp. 467–484, Aug. 1999.
- [205] G. C. Moore and I. Benbasat, "Development of an instrument to measure the perceptions of adopting an information technology innovation," *Inf. Syst. Res.*, vol. 2, no. 3, pp. 192–222, Sep. 1991.
- [206] S. Khemthong and L. M. Roberts, "Adoption of Internet and Web technology for hotel marketing: A study of hotels in thailand," J. Bus. Syst., Governance Ethics, vol. 1, no. 2, pp. 47–60, Jul. 2006.
- [207] T. L. Roberts, P. H. Cheney, P. D. Sweeney, and R. T. Hightower, "The effects of information technology project complexity on group interaction," *J. Manage. Inf. Syst.*, vol. 21, no. 3, pp. 223–247, Nov. 2004.
- [208] M. Khalifa and M. Davison, "SME adoption of IT: The case of electronic trading systems," *IEEE Trans. Eng. Manag.*, vol. 53, no. 2, pp. 275–284, May 2006, doi: 10.1109/tem.2006.872251.



SULFEEZA MOHD. DRUS received the bachelor's degree in management information system from Clarkson University, USA, in 1998, and the master's and Ph.D. degrees from Universiti Tenaga Nasional, in 2007 and 2014, respectively.

She is currently a Senior Lecturer with Universiti Tenaga Nasional. She is the Head of the Informatics Department, College of Computing and Informatics, Universiti Tenaga Nasional. Her research interests include, but are not limited to,

knowledge management, knowledge audit, and data science. Dr. Drus is a Dell EMC-Certified Data Science Associate.



NOR AZIZAH AHMAD received the Dip.Comp.Sc. and B.Com.Sc. degrees in network management from Universiti Putra Malaysia, and the master's degree in electronic business management from the International University of Japan, in 1999, 2002, and 2009, respectively. She is currently pursuing the Ph.D. degree in ICT with the Department of Informatics, College of Computing and Informatics, Universiti Tenaga Nasional, Putrajaya, Malaysia.

Her research interests include enterprise architecture adoption, electronic business management, and network management.



HAIROLADENAN KASIM received the bachelor's degree in information technology from Universiti Utara Malaysia, in 1998, and the master's degree and the Ph.D. degree in information management from Universiti Teknologi Mara, in 2007 and 2015, respectively.

He is currently working with the Department of Informatics, College of Computing and Informatics, Universiti Tenaga Nasional. He has conducted research on information systems (business infor-

matics); computing in social science, arts, and humanities; and computer and society. His most recent publication is Formulating a Digital Energy Hub Framework to Inculcate Knowledge Sharing Practices for Energy Sectors.

...